STREET LIGHT USE AGREEMENT

This Street Light Use Agreement ("Agreement") is made and entered into on this twenty-first day of February, 2006 by and between the Philadelphia Authority for Industrial Development ("PAID") and EarthLink, Inc., a Delaware Corporation ("EL" or "EarthLink"), with offices at 1375 Peachtree Street, Level A, Atlanta, GA 30309. The Agreement shall be effective as of the "Effective Date," as defined in Section 1, *Definitions*. PAID and EL may be referred to individually as a "Party" and collectively as the "Parties."

BACKGROUND

WHEREAS, extraordinary and unparalleled advances in technology relating to internet communication have had, and have the potential to expand, a profound effect on business and commerce, education and the lives of ordinary citizens;

WHEREAS, the ability to connect via the worldwide web offers accelerated access to information and the exchange of ideas, creating enormous opportunities for economic development, delivery of more efficient municipal services and for disadvantaged children and adults to realize their potential;

WHEREAS, the City has determined that the safety, general welfare, economic development and prosperity of the people of Philadelphia are dependent upon accelerated technological improvements to the communications systems available in the City and accessibility of those improvements to the broadest segments of the population;

WHEREAS, the City has determined to act as a catalyst to encourage, promote, and attract expanded communications systems within the City, and most particularly to develop a city-wide wireless network;

WHEREAS, the City has determined that this network alone will be insufficient to afford full internet accessibility to low income residents, but that programs focusing on hardware and software availability as well as education and training are also necessary;

WHEREAS, the City desires to foster and stimulate collaboration between and among private entrepreneurs, non-profit agencies and government to afford the broadest range of benefits of internet advances to the general public through a business model which will appropriately allocate the risks and rewards of city-wide internet deployment;

WHEREAS, the Philadelphia Authority for Industrial Development is a public instrumentality of the Commonwealth of Pennsylvania organized under the Pennsylvania Economic Development Financing Law, Act No. 102, approved April 23, 1967, as amended, for the purposes, among others, of promoting the installation and use of the most efficient means of

communications, promoting public and private infrastructure, and promoting industrial, commercial and other economic development in the City of Philadelphia;

WHEREAS, the City has determined that fostering the growth of a city-wide wireless network will be best accomplished by implementing a business model which will utilize municipally-owned assets for deployment and operation of such a network are made available to a locally-controlled not-for-profit corporation, Wireless Philadelphia;

WHEREAS, Wireless Philadelphia has determined that the implementation of such network will be best accomplished by partnering with a private entrepreneur;

WHEREAS, to support this business model, the City desires to grant to the Authority certain rights for the placement of wireless communications equipment on certain municipally-owned street light fixtures, referred to herein as the Street Lights, and to authorize the Authority to authorize EarthLink, a for-profit corporation with substantial business expertise as an internet service provider and a developer of municipal wireless networks, to use the Street Lights for the placement of such equipment as provided in this Agreement;

WHEREAS, in consideration for such right to use the Street Lights, the City will receive street light use fees, funding for certain costs, free and discounted service rights, and Wireless Philadelphia will receive portions of the revenues generated by EarthLink for the operation of its programs to lower the cost of internet access and to provide citizens of Philadelphia with the knowledge, training, and equipment to benefit from the area-wide network and to otherwise manage its operations;

WHEREAS, the Authority and EarthLink desire to enter into this Agreement to provide for EarthLink's right to use the Street Lights as provided herein, and the other terms and conditions regarding the city-wide network, upon approval by City Council of this Agreement and an underlying agreement between the City and the Authority.

NOW, THEREFORE, in consideration of the above referenced recitals and the following mutual covenants, agreements, and obligations of the Parties, which constitute good and valuable consideration, the sufficiency of which is acknowledged, and with the intention to be legally bound hereby, PAID and EL agree as follows:

1 Definitions.

Capitalized words in this Agreement shall have the meaning defined below and if not defined herein then the meaning defined in the Network Agreement.

- 1.1 "Accepted" is defined in Section 3.4(b) of the Network Agreement.
- 1.2 "Acceptance Criteria" is defined in Section 3.1(a) of the Network Agreement.
- 1.3 "Affiliates" is defined in Section 9.3.1.
- 1.4 "Agreement" is this Street Light Use Agreement between PAID and EarthLink unless specific reference to a different agreement is made.

- 1.5 "Alternate(s)" is defined in Section 3.2.2.
- 1.6 "Alternate Street Light(s)" is defined in Section 3.2.2.
- 1.7 "Alternative Service" is defined in Exhibit I Section 16 of the Network Agreement.
- 1.8 "Applicants" is defined in Section 3.1.1.
- 1.9 "Approved Updated Equipment" is defined in Section 2.4.2.
- 1.10 "As-Built Information" is defined in Section 4.2.1.
- 1.11 "Authorized Uses" is defined in Section 2.1.
- 1.12 "Bankruptcy Event" is defined in Section 7.4.
- 1.13 "Base Service" is defined in Exhibit A of the Network Agreement.
- 1.14 "Bulk Permit" is defined in Section 3.2.3.
- 1.15 "Bulk Construction Permit" means a Bulk Permit.
- 1.16 "Cable Franchise" is defined in Section 2.3.
- 1.17 "Cable Service" is defined in Section 2.3.
- 1.18 "City" means the City of Philadelphia, Pennsylvania.
- 1.19 "City/PAID Street Light Use Agreement" or "City/PAID Agreement" means the Street Light Use Agreement between PAID and the City of Philadelphia dated February 21, 2006 and signed February 21, 2006, pursuant to which the City approved this Agreement and consented to PAID's entering into this Agreement to authorize EL to use Street Lights as provided in this Agreement.
- 1.20 "Commissioner" or "Streets Commissioner" means the Commissioner of the Philadelphia Streets Department.
- 1.21 "Communications Equipment" means, collectively, the Tropos 5210 Radio and Motorola Canopy Radio and the Approved Updated Equipment.
- 1.22 "Confidential Information" is defined in Section 13.
- 1.23 "Construction Permit" means a permit issued by the City Department of Streets authorizing EL to install and maintain Communications Equipment on Street Lights as provided in Section 3.

- 1.24 "Coverage Area" is defined in Section 4.5 of the Network Agreement.
- 1.25 "Core ISP Services" are the Services set forth in Exhibit J of the Network Agreement.
- 1.26 "CPE" or "Customer Premises Equipment" means equipment that end users are required to provide in order to access the Wireless Network.
- 1.27 "City Council" means the City Council of Philadelphia.
- 1.28 "Dark" and "Dark Day" are defined in Section 12.1.3.
- 1.29 "Day," whether capitalized or not capitalized, means calendar day unless expressly provided otherwise.
- 1.30 "Design Concern Notice" is defined in Section 3.1(b) of the Network Agreement.
- 1.31 "Design Control Documents" is defined in Section 3.1(a) of the Network Agreement.
- 1.32 "Designated Street Light" is defined in Section 3.2.2.
- 1.33 "Digital Inclusion Areas" are those portions of the City identified on Exhibit E of the Network Agreement.
- 1.34 "Dispute" and "Dispute Resolution Period" are defined in Section 10.2.
- 1.35 "Earthlink" means EarthLink, Inc.
- 1.36 "EDPA" is defined in Section 17.19.
- 1.37 "Effective Date" means the later of (i) the date first written above, (ii) the date on which this Agreement and the City/PAID Agreement are fully executed, or (iii) the date on which an ordinance of City Council approving the City/PAID Agreement, to which this Agreement is attached as Exhibit A, becomes law.
- 1.38 "EL" means EarthLink.
- 1.39 "EL Assisted Wi-Fi" is defined in Exhibit I Section 3 of the Network Agreement.
- 1.40 "EL Retail Wi-Fi" is defined in Exhibit I Section 2 of the Network Agreement.
- 1.41 "EL Roaming Wi-Fi" is defined ins Exhibit I Section 4(d) of the Network Agreement.
- 1.42 "Final Acceptance" is defined in Section 3.4(d) of the Network Agreement.
- 1.43 "Finally Accepted" means the occurrence of Final Acceptance.
- 1.44 "First Payment Conditions" is defined in Section 4.1 of the Network Agreement.

- 1.45 "Force Majeure Events" is defined in Section 17.31.
- 1.46 "Future Products" are any access product offerings, in which the primary functionality shall be wireless broadband connectivity made available via the System, that are not set forth in Exhibit I of the Network Agreement (not including the reference to Future Products in Exhibit I of the Network Agreement).
- 1.47 "Implementation" is defined in Section 3.4 of the Network Agreement.
- 1.48 "Implementation Conditions" is defined is Section 4.3 of the Network Agreement.
- 1.49 "Indemnitees" is defined in Section 9.3.1.
- 1.50 "Initial Term" is defined in Section 11.1
- 1.51 "Motorola Canopy Radio" is defined in Section 2.4.1.
- 1.52 "Muni T1" is municipal T1 wireless service provided pursuant to Section 7.1.3.2.
- 1.53 "Muni Wi-Fi" is municipal broadband wireless service accounts provided pursuant to Section 7.1.3.1.
- 1.54 "Network" means the System as defined in the Network Agreement.
- 1.55 "Network Access Point" means the Motorola Canopy Access Point radio to which the Motorola Canopy Radio connects to create a wireless backhaul connection.
- 1.56 "Network Agreement" means the Wireless Philadelphia Broadband Network Agreement of even date herewith between Wireless Philadelphia and EarthLink, Inc.
- 1.57 "Network Construction Plan" is defined in Section 4.1.3.
- 1.58 "Network Operations" is defined in Section 9.3.1.
- 1.59 "Non-Termination Defaults" is defined in Section 12.3.
- 1.60 "OA Wi-Fi" is defined in Exhibit I Section 4 of the Network Agreement.
- 1.61 "OVS Franchise" is defined in Section 2.3.
- 1.62 "OVS Service" is defined in Section 2.3.
- 1.63 "PAID" or the "Authority" means the Philadelphia Authority for Industrial Development.
- 1.64 "Party" means either of the PAID or EarthLink.

- 1.65 "Parties" means both PAID and EarthLink.
- 1.66 "PECO" means Exelon Corporation.
- 1.67 "Permittee(s)" is defined in Section 3.1.1.
- 1.68 "Permits" means Construction Permits.
- 1.69 "Person" means any individual natural person, firm, partnership, joint venture, entity, society, organization, club, association, trustee, trust, corporation, company or organization of any kind.
- 1.70 "PIDC" means the Philadelphia Industrial Development Corporation, a nonprofit corporation organized under the laws of Pennsylvania for the purpose, among others, of assisting PAID in carrying out PAID's economic development and other purposes.
- 1.71 "POC Construction Plan" is defined in Section 4.1.2.
- 1.72 "Preliminary Design Review" is defined in Section 3.1(a) of the Network Agreement.
- 1.73 "Proof of Concept Acceptance" is defined is Section 3.2 (c) of the Network Agreement.
- 1.74 "Proof of Concept Conditions" is defined in Section 4.2 of the Network Agreement.
- 1.75 "Proof of Concept Trial" is defined in Section 3.2(d) of the Network Agreement.
- 1.76 "Proof of Concept Permits" is defined in Section 4.2(b) of the Network Agreement.
- 1.77 "Renewal Terms" is defined in Section 11.2.
- 1.78 "Request for Internal Resolution" is defined in Section 10.2.
- 1.79 "Revenue Share" means the payments made to WP as provided in the Network Agreement.
- 1.80 "Right-of-Way" has the meanings provided in Chapter 11-700 of The Philadelphia Code.
- 1.81 "Right-of-Way Use License" has the meaning provided in Chapter 11-700 of The Philadelphia Code.
- 1.82 "Schedule" is defined in Section 2 of the Network Agreement.
- 1.83 "Streets Department" means the Department of Streets of the City.
- 1.84 Reserved.
- 1.85 "Street Light" means the City-owned street light fixture and attached photocell, together with the lateral arm, overhanging the street and/or sidewalk, on which the light fixture is

- mounted. "Street Light" does not mean or include any pole (whether or not owned by the City) to which the lateral arm and light fixture are attached.
- 1.86 "Streets Commissioner" means the Commissioner of the Department of Streets of the City.
- 1.87 "Subcontractor" means any contractor of EL and/or of any EL contractor, at any tier, that performs work on the Wireless Network, whether or not EL is directly in contract therewith.
- 1.88 "System" means the Wireless Network.
- 1.89 "System Acceptance" in defined in Section 3.4(e) of the Network Agreement.
- 1.90 "T1 Alternative Product" is defined Exhibit I of the Network Agreement.
- 1.91 "Term" and "term of this Agreement" mean the Initial Term and any Renewal Terms.
- 1.92 "Termination Default(s)" is defined in Section 12.1.
- 1.93 "Three Important SLAs" are items 1, 2 and 5 set forth in Exhibit F under "Performance Requirements" of the Network Agreement.
- 1.94 "Three Important SLAs Failure Notice" is defined in Section 16.4(a) of the Network Agreement.
- 1.95 "Tropos 5210 Radio" is defined in Section 2.4.1.
- 1.96 "Wi-Fi Broadband Product" is defined Exhibit I of the Network Agreement.
- 1.97 "Wireless Network" or "System" means the City wide wireless broadband Internet access system that EL will design, build, install, test, operate, maintain and upgrade as provided in the Network Agreement; the Wireless Network to be in accordance with Exhibit B, *Network Description*, hereto and Exhibit F, *Requirements*, to the Network Agreement, as such Exhibits may be amended pursuant to the Network Agreement.
- 1.98 "WP" means Wireless Philadelphia.
- 1.99 "WP Mission" is defined in Section 1.5 of the Network Agreement.
- 1.100 "Zone" is defined in Section 3.4(a) of the Network Agreement.
- 2 Authorization.
- 2.1 <u>Scope of Authorization</u>. Subject to the terms and conditions of this Agreement and for the term provided in Section 11, *Term, Renewal, Expiration and Removal*, PAID authorizes EL to install, repair, replace, operate, maintain, and use Communications Equipment on the Street Lights for which the City has issued Construction Permits in accordance with

Section 3, Application and Approval to Place Communications Equipment (collectively, the "Authorized Uses"). Neither EL nor its Subcontractors may use or perform any activity on any Street Light without first obtaining a Construction Permit for that Street Light in accordance with Section 3, and nothing in this Section 2 or elsewhere in this Agreement shall be construed to limit or excuse such requirement to obtain a Construction Permit. EL's authorization to use Street Lights shall terminate upon the termination of this Agreement in accordance with its terms. Nothing in this Agreement shall authorize or convey any right to EL to install Communications Equipment or other equipment on or otherwise to use any pole to which a Street Light is attached, whether or not the pole is owned by the City.

- 2.2 <u>Authorization Not Exclusive</u>. The authorizations granted in this Section 2 shall be non-exclusive and nothing contained in this Agreement shall be construed to limit, alter, or waive the right of PAID or the City to enter into agreements authorizing persons or entities other than EL to access and use Street Lights to install any form of communications equipment or for any other purpose. Such authorizations by the City or PAIDshall be made on a non-discriminatory and competitively neutral basis, taking into account material differences in the circumstances of each applicant for such authorization.
- 2.3 <u>Cable Service and OVS Service</u>. EL shall not provide Cable Service or OVS Service by means of the Wireless Network without first obtaining a Cable Franchise or OVS Franchise as required by Section 11-703, *Cable Franchise; Open Video System Franchise*, of the Philadelphia Code, as amended; such obligation is subject to the enforceability of Section 11-703 under applicable law. EL shall obligate its wholesale customers to comply with the foregoing sentence, shall enforce the obligation, and shall incorporate in its service agreements with such customers a provision substantially the same as this Section 2.3. The terms "Cable Service," "OVS Service," "Cable Franchise," and "OVS Franchise," as used in this Section 2.3 and elsewhere in the Agreement, shall have the meanings provided for them in Section 11-701, *Definitions*, of The Philadelphia Code.
- 2.4 Authorized Communications Equipment and Installation Method; Upgrades.
- 2.4.1 EL is authorized to place on Street Lights the following Communications Equipment:

 The Tropos Model 5210 Outdoor MetroMesh Router radio device depicted in Exhibit A
 (the "Tropos 5210 Radio") and the three Motorola Canopy Broadband Wireless
 Subscriber Module radio devices depicted in Exhibit A (collectively, the "Motorola
 Canopy Radio"). The method of installation shall be as depicted in Exhibits A. In no
 event may any part of the Street Light be penetrated by any radio device, cable, mounting
 device, or other Communications Equipment. Electrical power shall be obtained only by
 attaching an adapter and cable to the Street Light photo cell, in the manner depicted in
 Exhibit A. Except as authorized by the Streets Commissioner in writing, no other item of
 Communications Equipment (including without limitation upgraded or updated versions
 of Communications Equipment) or other equipment or material may be placed on any
 Street Light; no other method of installation may be employed; and no other means of
 obtaining electrical power for Communications Equipment installed on the Street Lights
 may be used.

- 2.4.2 Authorization of the Streets Commissioner to replace authorized Communications Equipment with updated or upgraded Communications Equipment shall be required prior to installation but will not be unreasonably withheld provided the replacement equipment (i) is no heavier or larger and uses no more electrical power than the Tropos 5210 Radio and/or Motorola Canopy Radio(s), and uses the same method of obtaining electrical power as those two radio devices; and (ii) the electrical power consumption, size and weight of the updated equipment does not, in the reasonable judgment of the Streets Commissioner, place an unacceptable burden or risk on the Street Lights (equipment satisfying all of the foregoing requirements is referred to herein as "Approved Updated Equipment"). The installation of Approved Updated Equipment in substitution of Communications Equipment for which Construction Permits have been issued by the Streets Commissioner shall not normally require the issuance of new Construction Permits provided the Approved Updated Equipment is placed on a Street Light for which a Construction Permit has been issued. Notwithstanding the foregoing, such installations shall be removed or altered as directed by the Streets Commissioner if, upon inspection, they are determined by the Streets Commissioner to be noncompliant with the requirements of the Construction Permit originally issued or the requirements of this Agreement. Software upgrades to the Wireless Network that do not require replacement or modification of any Communications Equipment apart from the installation of the software remotely through the Wireless Network may be performed without prior approval of the Commissioner; provided, however, that if the Streets Commissioner requests notifications prior to software upgrades, EL shall provide the requested notification.
- 2.4.3 Except as authorized by the Streets Commissioner in writing, not more than one (1) Tropos 5210 Radio and not more than one Motorola Canopy Radio may be attached to any Street Light; provided, however, that different radio devices may be substituted for these devices as part of an upgrade of the Communications Equipment in accordance with Sections 2.4.2 and 2.4.4. The City will not, subsequent to issuing EL a Construction Permit for its use of a Street Light, authorize other persons or entities to place radio devices on the Street Light if the Streets Commissioner reasonably determines that such additional radio device(s) will cause radio frequency interference or physical interference with the operation of the EL radio device(s) for which a Construction Permit was issued.
- 2.4.4 In no event may EL or any of its Subcontractors install or construct new Street Lights or modify or repair existing Street Lights except as may be expressly authorized by this Agreement or by a Construction Permit issued pursuant to Section 3, or otherwise authorized in writing by the Streets Commissioner.
- 2.4.5 In no event may EL or any of its Subcontractor(s) attach any Communications Equipment or any other device or material to a Street Light, or modify the location or method of attachment, except as expressly authorized by this Agreement or by a Construction Permit issued pursuant to Section 3, *Application and Approval to Place Communications Equipment*, or is otherwise authorized in writing by the Streets Commissioner. Upgrades of Communications Equipment that require the replacement of the equipment or any appurtenant equipment shall not be made unless authorized in writing by the

- Commissioner in accordance with Section 2.4.1, and, if deemed reasonably necessary by the Streets Commissioner, a Construction Permit pursuant to Section 3 is issued therefor.
- 2.5 Conditions of City Issuance of Construction Permits. The City's issuance of new Construction Permits to EL under this Agreement is conditioned on EL's compliance with the following Sections 2.5.1-2.5.7 at all times during the Initial Term and any Renewal Terms of this Agreement. EL's failure to comply with such conditions shall constitute a Non-Termination Default under Section 12, *Default and Termination*, and shall entitle the City to exercise the remedies set forth therein, including, without limitation, the remedy of denying new applications for Construction Permits set forth in Section 12.2; provided, however, that Construction Permits will not be denied except for EL's persistent failure to comply (as reasonably determined by the Streets Commissioner) after written notice of noncompliance by the Streets Department; and provided further that EL may, after taking appropriate action to ensure future compliance, apply to the Streets Commissioner for reinstatement of the issuance of Construction Permits.
- 2.5.1 EL shall comply with the terms of this Agreement.
- 2.5.2 The Streets Commissioner shall have issued a Right-of-Way Use License, in accordance with Chapter 11-700 of The Philadelphia Code, to EL, as the owner of Communications Equipment placed on Street Lights, and/or to each of EL's Subcontractor(s) that owns or will own Communications Equipment placed on the Street Lights.
- 2.5.3 EL shall obtain, prior to commencing installation or operation, all applicable permits, licenses, and authorizations from the City, Commonwealth of Pennsylvania, and the federal government for the installation and operation of the Wireless Network and such authorizations have not been terminated or revoked.
- 2.5.4 EL and its Subcontractors shall comply with Section 2.14, Compliance With Law.
- 2.5.5 Reserved.
- 2.5.6 EL shall pay to PAID for the account of the City and to WP, as applicable, all compensation required under Section 7, *Compensation*, of this Agreement and/or under the Network Agreement when due in accordance with the terms thereof.
- 2.5.7 The conditions provided in Section 2.7, *Conditions of the Obligations of the Parties*, are satisfied and neither party has terminated the Agreement pursuant to Section 2.7.7, *Termination*.
- 2.6 <u>Maximum Number of Street Lights Authorized</u>. EL shall be authorized to access and to use, pursuant to this Agreement, a total of not more than five thousand (5,000) Street Lights throughout the City and shall not access or use more than that number of Street Lights; provided, that the Streets Commissioner, in her sole discretion and upon application by EL, may authorize EL's use of additional Street Lights in excess of five thousand (5,000).

- 2.7 <u>Conditions of the Obligations of the Parties</u>. The obligations of PAID and EL hereunder shall be subject to the following conditions:
- 2.7.1 This Agreement has been approved and fully executed by all Parties through fully authorized representatives.
- 2.7.2 The Philadelphia City Council has approved by ordinance such agreements between the City and PAID as may be required to authorize PAID to enter into this Agreement and give to EL the rights to use Street Lights that are provided in this Agreement, and such ordinance has become law.
- 2.7.3 PECO will furnish electricity service to the Communications Equipment installed on Street Lights and in a manner that the Parties agree complies with applicable laws and regulations.
- 2.7.4 EL is not prohibited from constructing and operating the Wireless Network, or performing substantially all of the services or exercising substantially all of the rights set forth in this Agreement by the final decision, following exhaustion of all appeals, of a court of competent jurisdiction, or by the enactment into law of a valid and enforceable federal, state, or local statute.
- 2.7.5 EL has obtained all applicable permits, licenses, and authorizations from the City, Commonwealth of Pennsylvania, and the federal government for the installation and operation of the Wireless Network, and such authorizations have not been terminated or revoked and are no longer subject to appeal; provided, that EL may not, on account of the failure of this condition to be satisfied, terminate the Agreement under Section 2.7.7 unless EL has exercised every commercially reasonable effort to obtain all such permits, licenses, and authorizations; and PAID may not so terminate the Agreement if the City has unreasonably withheld such authorizations that are required from the City and within the City's control to issue.
- 2.7.6 Proof of Concept Acceptance has occurred pursuant to Section 3.2(c) of the Network Agreement, and the City and WP has each issued its written acceptance of the Proof of Concept in accordance with the procedures provided in Section 3.2(c) of the Network Agreement. The City shall have the right to participate in the Proof of Concept Acceptance process as provided in Section 3.2(c) of the Network Agreement.

2.7.7 <u>Termination</u>.

2.7.7.1 Either Party shall be entitled, but not obligated, to terminate this Agreement without penalty or liability to the other Party if, on the day that is one hundred and eighty (180) days after the later of (i) the date on which the ordinance of City Council approving the City/PAID Agreement to which this Agreement is attached as Exhibit A has become law, (ii) the Effective Date, or (iii) the date on which the Agreement is fully executed, the conditions set forth in Sections 2.7.1-2.7.6 are not completed and satisfied, and such failure to complete and satisfy was not caused by the terminating Party; provided, however, as follows: a party shall terminate under this Section by delivering its written notice of termination to the other Party and to the City within

ninety (90) days following the end of such one hundred and eighty (180) day period; the Agreement shall remain in full force and effect throughout the ninety (90) day notice period and shall terminate on the last day of the notice period unless such conditions are completed and satisfied by the end of the ninety (90) day notice period, or the Agreement is sooner terminated by the written agreement of both parties or in accordance with its terms.

- 2.7.7.2 In addition to the foregoing right of termination, either party shall be entitled, but not obligated, to terminate this Agreement without penalty or liability to the other Party if at any time during the Initial Term or any Renewal Terms of the Agreement, EL is prohibited from constructing and operating the Wireless Network, or the City, PAID, or EL is prohibited from performing substantially all of the obligations or exercising substantially all of the rights set forth in this Agreement, by the final decision, following exhaustion of all appeals, of a court of competent jurisdiction, or by the enactment into law of a valid and enforceable federal, state, or local statute. A Party that believes it has the right to terminate under this Section 2.7.7.2 may elect not to terminate without waiving its right to terminate. In the event of termination pursuant to this Section, EL shall remove the Communications Equipment and restore the Street Lights as provided in Section 11.3, *Expiration and Removal*, and the City shall have all the rights and remedies provided therein.
- 2.7.7.3 Notwithstanding anything to the contrary in this Section 2.7.7., (i) if the Network Agreement is terminated because Proof of Concept Acceptance is not achieved within the ninety (90) day period provided in Section 3.2(d) of the Network Agreement, then this Agreement shall terminate on the date the Network Agreement so terminates; and (ii) if City Council has not approved this Agreement and the City/PAID Street Light Use Agreement by ordinance (each in the form and content executed by all parties, dated February 21, 2006 and effective the Effective Date of this Agreement, and attached to the proposed ordinance as introduced in City Council), and/or the ordinance has not become law on or before the earlier of (x) the end of the Spring 2006 Council term or (y) July 1, 2006, then each Party shall have the right to terminate this Agreement with ten (10) days written notice to the other Party, with a copy to the City; provided, that such approval of this Agreement shall be the approval of the City/PAID Street Light Use Agreement with this Agreement attached as Exhibit A thereto. If a Party elects not to terminate under this Section 2.7.7.3 at the time it is entitled to do so, it shall not have waived its right to later terminate.
- 2.8 Except as expressly provided otherwise in this Agreement, nothing in the Agreement, or in any ordinance of City Council approving this Agreement, shall constitute an authorization to install, operate, maintain, repair, modify, or use Communications Equipment or any other device or material on a Street Light without first obtaining a Construction Permit as provided herein.
- 2.9 Nothing in this Agreement shall constitute a representation or warranty that the City or PAID has sought or obtained permission from any third-party property owner for EL's installation or operation of Communications Equipment or any other equipment on that third party's property unless the Communications Equipment is on Street Lights located in

- the Right-of-Way. EL acknowledges and agrees that it shall be solely responsible for obtaining any such permission that may be required for its installation or operation of Communications Equipment that is not placed on Street Lights located in the Right-of-Way.
- 2.10 Neither EL's occupancy nor its use of Street Lights or the Right-of-Way, nor anything contained in this Agreement or in any Construction Permit issued by the City (i) shall grant, convey, create, or vest in EL any form of real property interest in any Street Light or other City personal property or real property or facility, or in any pole to which a Street Light is attached (whether or not owned by the City), or in any portion of the Right-of-Way, including, but not limited to, any fee or leasehold interest, any easement, or any franchise right therein; or (ii) create in PAID or the City any ownership interest in any Communications Equipment installed by EL on a Street Light, except as may be expressly provided in this Agreement.
- 2.11 EL acknowledges and agrees that neither PAID nor the City makes and has made no warranties or representations regarding the fitness, safety or suitability of the Street Lights or any City property or facility, or the Right-of-Way, for the installation or operation of the Communications Equipment or the Wireless Network; that EL is solely responsible for determining such fitness, safety, and suitability; and that any performance of work, costs incurred, and/or provision of services contemplated under this Agreement by EL is at EL's sole risk.
- 2.12 Wireless Internet access shall at all times throughout the Initial Term and any Renewal terms be offered over the Wireless Network; such access will be provided in the form of "Base Service" as defined and set forth in Exhibits A and I of the Network Agreement.
- 2.13 EL shall construct and operate the Wireless Network in accordance with the Network Agreement.
- 2.14 Compliance With Law. EL and its Subcontractors shall comply with all applicable federal, state, and local laws, ordinances and regulations, including, without limitation, Chapters 9-300, 11-200, and 11-700 of The Philadelphia Code, relating to management of the Right-of-Way; all other applicable provisions of The Philadelphia Code, and the Regulations of the Department of Streets and the Department of Public Property. Nothing contained in this Agreement shall limit or waive EL's obligation to comply with ordinances of general application of the City and provisions of general application of The Philadelphia Code that become law after the date first written above.
- 3 Application and Approval to Place Communications Equipment.
- 3.1 Construction Permits Required.
- 3.1.1 Neither EL nor its Subcontractors may place any Communications Equipment on any Street Light or use the Street Lights for any Authorized Use unless or until EL has applied for and the Streets Commissioner has issued a Construction Permit authorizing

- the activity on that Street Light. (Parties applying for such Construction Permits are referred to in this Agreement as "Applicants" and recipients as "Permittee(s).")
- 3.1.2 EL shall not, however, be required to obtain a Construction Permit to perform routine maintenance on Communications Equipment, or to replace an item of Communications Equipment with the same model of equipment provided (i) the model has been approved in writing by the Streets Commissioner in this Agreement or otherwise, (ii) the replacement device is no heavier or larger and uses no more electrical power than the model it is replacing, (iii) the method of obtaining electrical power for the device is the same as for the model it is replacing, and (iv) the electrical power consumption, size and weight of the updated equipment does not, in the reasonable judgment of the Streets Commissioner, place an unacceptable burden or risk on the Street Lights.
- 3.1.3 If EL has not installed Communications Equipment on a Street Light within one hundred and eighty (180) days after the issuance of a Construction Permit for the Street Light (or a longer period if provided in the Construction Permit), the Construction Permit shall be deemed expired as to that Street Light, and EL's right to use the Street Light shall be terminated. EL may, however, apply for a new Construction Permit for any Street Light as to which a Construction Permit has expired and its use right terminated pursuant to this Section.

3.2 Permit Applications.

- 3.2.1 <u>Information Required</u>. Applications for Construction Permits shall contain the following information and plans, and such additional information as the Streets Commissioner may reasonably require in order to evaluate the proposed construction:
 - The exact location of each Street Light on which the Applicant seeks to place
 Communications Equipment, identified by the pole identification number assigned to
 the pole in the Streets Department geographical information system database. The
 Streets Department will provide EL with sufficient information to comply with this
 requirement.
 - Detailed specifications for each item of equipment to be placed on the Street Light, showing size, weight, mounting method, method of providing electrical power (including placement of any cables), and the method of attaching to the Street Light photo cell if required for electrical power.
- 3.2.2 <u>Substitution</u>. EL may apply for Construction Permits for more Street Lights than it intends to use for the placement of Communications Equipment in any block or blocks of the City, by designating, in its application, the intended Street Light (the "Designated Street Light") and up to two alternate Street Lights (the "Alternate Street Light(s)" or "Alternate(s)"). If such permits are issued, the Permittee may substitute any one of the Alternates approved in the permit for the Designated Street Light, if substitution is necessary for proper coverage and operation of the Communications Equipment; but EL shall be deemed to have waived its right to use any of the permitted Street Lights on which it has not placed Communications Equipment within ninety (90) days of the

issuance of the permit. Within the times following installation provided in Section 4.2.2 (relating to submission of As-Built Information during initial construction), EL shall submit to the City as-built information, in form and content determined by the Streets Commissioner, that identifies the specific Street Light, Designated or Alternate, on which the Communications Equipment was installed. If authorized by the Streets Commissioner, substitutions of alternate Street Lights for Designated Street Lights may be approved in the field by authorized Streets Department inspectors whether or not the alternate Street Lights were approved in the permit.

- 3.2.3 <u>Bulk Construction Permit Applications</u>. EL may apply for a single Construction Permit that applies to up to one hundred and twenty-five (125) Designated Street Lights, plus up to two Alternates for each (a "Bulk Permit"); provided, that EL shall not apply for and the Streets Department shall not be obligated to process more than one Bulk Permit application per week without the written authorization of the Streets Commissioner. The application for a Bulk Permit must comply with all requirements of this Section 3. Section 3.2.2, *Substitution*, shall apply to Bulk Permits.
- 3.2.4 <u>Time to Issue</u>. For applications for up to ten (10) Street Lights, the Streets Department shall exercise reasonable efforts to issue Construction Permits or its rejection of the permit application within fourteen (14) days of receiving a complete application. For applications for Bulk Permits, the Streets Department shall exercise reasonable efforts to issue Construction Permits or its rejection of the application within thirty (30) days of receiving a complete application. Applications will not be unreasonably rejected.
- 3.2.5 <u>Lane Closure Permits</u>. EL or its Subcontractor(s) shall not close or block vehicular traffic in any traffic lane of a City street without first obtaining a lane closure permit or other authorization as required by the Streets Department.
- 4 Network Construction; Construction Plans.
- 4.1 Construction Plans; Organization of the Work.
- 4.1.1 Organization of the Work. EL shall schedule the work so that each construction crew completes the installation of Communications Equipment on all Street Lights authorized by a Bulk Construction Permit before that crew commences the installation of Communications Equipment on Street Lights authorized by a different Bulk Construction Permit. No construction crew shall at any time perform installations on Street Lights covered by two or more Bulk Construction Permit. Nothing in this Section 4.1.1 shall preclude EL from requesting or the Streets Commissioner from approving a different organization of the work.
- 4.1.2 <u>Proof of Concept Construction Plan</u>. EL shall submit to the Streets Commissioner its detailed plan for the construction of the Proof of Concept Trial ("POC Construction Plan") at least thirty (30) days prior to commencing construction of the network for the Proof of Concept Trial. The network for the Proof of Concept Trial shall cover a total area of approximately fifteen (15) square miles and shall be constructed as provided in the Network Agreement. The POC Construction Plan shall include the following

information, together with such additional information and plans as the Streets Commissioner may reasonably require:

- A detailed map, in electronic form and in a format reasonably required by the Streets
 Commissioner, depicting (i) each block of the City in which EL and its
 Subcontractor(s) intend to place Communications Equipment on Street Lights; the
 number of Street Lights in each block for which use is intended; and the locations
 (including whether a roof top or tower) for each Network Access Point to be
 constructed.
- A construction schedule depicting start date, completion date, and the sequence of construction for the POC Pilot area.

The POC Construction Plan, including without limitation the map and schedule, shall be updated and the updated Plan submitted to the Streets Commissioner at least every thirty (30) days throughout the construction of the POC Pilot network.

- 4.1.3 Wireless Network Construction Plan. EL shall submit to the Streets Commissioner its detailed plan for construction of the entire Wireless Network (the "Network Construction Plan") at least thirty (30) days prior to commencing construction for any part of the Wireless Network other than the Proof of Concept Trial (and any earlier constructed pilots authorized by the Streets Commissioner). The Network Construction Plan shall divide the geographical area of the City into at least five (5) Zones, approximately equal in geographical area and each consisting of contiguous City blocks. The Network Construction Plan shall include the following information, together with such additional information and plans as the Streets Commissioner may reasonably require:
 - A detailed map, in electronic form and in a format reasonably required by the Streets
 Commissioner, depicting (i) each block of the City in which EL and its
 Subcontractor(s) intend to place Communications Devices on Street Lights; the
 number of Street Lights in each block for which use is intended; and the locations
 (including whether a roof top or tower) for each Network Access Point to be
 constructed.
 - A construction schedule depicting the start and completion times for construction in each Zone, and the sequence of construction within each Zone.

The Network Construction Plan, including without limitation the map and schedule, shall be updated, and the updated Plan submitted to the Streets Commissioner, within thirty (30) days following completion of the Proof of Concept Trial, and every thirty (30) calendar days thereafter until construction of the entire Wireless Network is completed and Finally Accepted in accordance with the Network Agreement.

4.2 As-Built Information.

4.2.1 EL shall maintain and submit to the Streets Commissioner accurate, fully up-to-date asbuilt plans, maps, drawings and information identifying all Street Lights and other City

property and facilities in use by EL, including the Communications Equipment and any other equipment that is located on the Street Light or other property or facility owned or controlled by the City (the "As-Built Information"). In addition to the foregoing, the As-Built Information shall include, for each Street Light on which Communications Equipment is placed: the pole identification number assigned to the pole and Street Light in the Streets Department geographical information system database (the Streets Department will provide EL with sufficient information to comply with this requirement); the serial number, manufacturer, model number and device type (short description) of each item of equipment placed on the Street Light; the electrical power source; and the contact information required on the information tags provided in Section 5.9, *Information Tags*; and such additional information as the Streets Commissioner determines is necessary to manage EL's use of the Street Lights and the Right-of-Way. The As-Built Information shall be submitted in a format determined by the Streets Commissioner.

- 4.2.2 During the period of initial construction of the Wireless Network (i.e. the period from execution of the Network Agreement through Final Acceptance of the Wireless Network in accordance with the Network Agreement), EL shall submit the As-Built Information as follows: (i) On or before the Friday of each week, EL shall submit to the Streets Commissioner a map or other form of information approved by the Streets Commissioner that accurately identifies each Street Light on which EL placed Communications Equipment during the prior week, together with a description of the Communications Equipment installed; (ii) within ten (10) days following its completion of the installation of Communications Equipment in a Zone, EL shall submit to the Streets Commissioner As-Built Information (other than maps, plans and drawings) for the entire Zone, and shall submit maps, plans and drawings required as part of the As-Built Information within thirty (30) days following such completion.
- 4.2.3 Following Final Acceptance of the Wireless Network and throughout the term of this Agreement and any renewal term, EL shall submit revised and updated As-Built Information as follows for new installations of Communications Equipment and for relocations of existing installations to different Street Lights: (i) Within thirty (30) days following Final Acceptance, EL shall submit to the Streets Commissioner a map (or other form of information approved by the Streets Commissioner) that accurately identifies each Street Light on which EL placed Communications Equipment; (ii) within ten (10) days following any change from the last-submitted As-Built Information for a Zone, EL shall submit to the Streets Commissioner updated As-Built Information (other than maps, plans and drawings) for the entire Zone, and shall submit maps, plans and drawings required as part of the As-Built Information within thirty (30) days following such completion.

- 5 Installation And Maintenance.
- 5.1 <u>Inspection</u>.
- 5.1.1 <u>Construction Inspection</u>.
- 5.1.1.1 All construction activities relating to Street Lights and/or taking place in the Right-of-Way, including, without limitation, the installation of Communications Equipment on Street Lights, shall be subject to such inspections as the Streets Commissioner determines are necessary to ensure the construction complies with this Agreement, Streets Department regulations, and applicable provisions of The Philadelphia Code. In the event Streets Department inspection determines that such construction activities are noncompliant, EL shall promptly correct the noncompliance as directed by the Streets Department.
- 5.1.1.2 The Streets Department shall provide three dedicated full-time construction inspectors to accompany up to three EL installation crews throughout the construction of the Wireless Network, up to a maximum of twelve (12) months, subject to its ability to identify and retain the three inspectors in accordance with the City's hiring requirements. EL shall pay all Streets Department costs for the three inspectors and their inspection activities, up to a maximum of Three Hundred Thousand Dollars (\$300,000). The Streets Department will submit invoices to EL for such inspection costs periodically, in a manner and with such documentation as is reasonably determined by the Streets Commissioner and EL. If more than three EL construction crews operate concurrently, or if the Streets Department is unable to identify and retain the three inspectors in accordance with City hiring requirements, EL shall be responsible for providing inspection services, through a third party contractor approved by the Streets Commissioner and in accordance with Streets Department inspection requirements, at EL's cost.
- 5.1.2 <u>Safety Inspections</u>. The Streets Department shall have the right to conduct such inspections of the Communications Equipment and other EL facilities and equipment located on Street Lights, at other City-owned or City-controlled sites or facilities, and/or in the Right-of-Way, as the City deems reasonably necessary in order to ensure the safety thereof, compliance with this Agreement and applicable Construction Permits, and the health, safety, and welfare of the public, including, without limitation, periodic safety inspections by the Streets Department of Communications Equipment located on Street Lights. The City shall have full access to the Communications Equipment.
- 5.2 <u>Removal of Previous Installations</u>. If, during installation of Communications Equipment on a Street Light, EL encounters a device previously installed by others, it may apply to the Streets Commissioner for authority to remove the device. In no event shall any such device be removed without the written approval of the Streets Commissioner, and in no event shall EL or its Subcontractors remove any such device that is operating and in use by its owner.

- 5.3 Removal for City Work. The City shall have the right to require EL to temporarily or permanently remove or relocate its Communications Equipment and/or any other equipment located on Street Lights or other City property if the removal or relocation is necessary to accommodate the construction, completion, repair, relocation or maintenance of a City project, or in the event the City replaces, repairs, or alters a Street Light on which EL has placed Communications Equipment. The Streets Commissioner shall notify EL thirty (30) days before the date on which removal or relocation is required to commence, and shall provide a longer notice period where the Commissioner reasonably determines that a longer notice period is necessary or appropriate for the work required. EL shall complete such removal or relocation, at EL's sole cost and expense, after written notice from the Streets Department and within a period of time reasonably determined by the Streets Commissioner, which shall be at least thirty (30) days. If EL does not comply with this Section 5.3, the City may remove or relocate the Communications Equipment or other equipment at EL's expense. If EL is directed, under this Section 5.3, to remove its Communications Equipment from a Street Light it is authorized to use, EL may elect to relocate, replace, or transfer some or all of the Communications Equipment to replacement Street Lights, at EL's cost and subject to the Streets Commissioner issuing Construction Permits, if required by the Commissioner, for the proposed replacement Street Light(s).
- 5.4 Emergency Action. The Streets Commissioner shall have the right to require the immediate removal, relocation, replacement, or transfer of Communications Equipment in the event the Commissioner determines that an emergency condition, as reasonably determined by the Commissioner, requires such action. In such event, the Commissioner shall promptly inform EL that such action is required and EL shall immediately complete the action required by the Commissioner. If the Commissioner determines there is a threat to the public health, safety, or welfare, the Streets Department shall have the right to remove, relocate, replace, or transfer the Communications Equipment at EL's expense. Absent such threat, the Streets Department shall not itself relocate, replace, or transfer the Communications Equipment without EL's prior written consent.
- 5.5 Damaged or Destroyed Street Lights.
- 5.5.1 If a Street Light or Street Light pole is damaged or destroyed to such an extent that EL or its Subcontractors cannot safely or lawfully maintain its operations at its then current level using that Street Light or pole, then the Party that first becomes aware of the problem shall provide the other Party, the Streets Commissioner, and PECO, if the condition affects a PECO owned pole or the electrical power supply, with written notice of this circumstance within a reasonable period after the damage, destruction, or inability to maintain operations. The Streets Commissioner shall then have thirty (30) days to advise EL in writing as to whether the City intends to restore or replace the applicable Street Light or pole, if the pole is owned by the City, and how long such replacement will take. If the pole is owned by PECO, the Commissioner shall exercise reasonable efforts to determine whether PECO intends to restore or replace the pole and will notify EL in writing of PECO's intention within thirty (30) days of PECO's notice to the Commissioner as to its intention. If the Streets Department advises EL that neither it nor PECO intends to restore or replace the affected Street Light or pole within thirty (30) days from the Commissioner's notice, EL shall have the right to designate a substitute

Street Light for the affected Communications Equipment and request, in writing, the Commissioner's approval to relocate its equipment to the designated Street Light. Within thirty (30) days after its receipt of EL's written request, the Streets Department shall notify EL whether it will permit EL to relocate its Communications Equipment to the substitute Street Light. Provided the requested Street Light(s) are not being used by others and satisfy the requirements of this Agreement, the Streets Department's approval will not be unreasonably withheld. EL shall be solely responsible for the cost of transferring Communications Equipment to the substitute pole and any costs required to make the pole ready for installation.

- 5.5.2 If the Streets Department elects not to repair or replace the destroyed or damaged Street Light or (where owned by the City) pole within thirty (30) days, then EL's right to use the Street Light and pole shall terminate, effective as of the date the Street Light and/or pole became unusable for the Communications Equipment. EL shall be entitled to a prorata abatement of the fee provided in Section 7.1.2, *Monthly Street Light Use Fee* for the affected Street Light for the number of twenty-four (24) hour periods it is unable to conduct its normal operations from the affected Street Light or the substitute Street Light, but shall otherwise be obligated to pay the fee provided in Section 7.1.2 for each such period that it uses or has access to a Street Light for the installation of Communications Equipment.
- 5.5.3 If the Streets Department elects to repair or replace the original Street Light or Street Light pole after the deadlines specified above have expired, then it shall provide notice thereof to EL. Subject to the Streets Commissioner's approval of its use of the original Street Light or its continued use of the substitute Street Light, whichever is applicable, EL shall then have thirty (30) days after the restoration or replacement has been completed to either (i) transfer the Communications Equipment back to the original Street Light and reinstate this Agreement as to the original Street Light; or (ii) notify the Commissioner that it will continue to maintain the Communications Equipment on the substitute Street Light pursuant to this Agreement.
- 5.5.4 No Obligation to Replace, Construct or Improve Street Lights. Neither PAID nor the City shall have any obligation to construct new Street Lights (including poles and appurtenant equipment), or to replace, repair, modify, upgrade, or improve any Street Light (including poles and appurtenant equipment).
- 5.5.5 Restoration. EL shall, at its own expense, promptly restore to its original condition or a better condition, in accordance with City regulations and standards generally applicable to such work, any street or Right-of-Way affected by any work authorized under this Agreement and/or any Construction Permit issued by the City, including both the sidewalk and roadway, and any street light fixture, street light pole, traffic signal pole or other City property affected, all as reasonably required by the Streets Commissioner. During the construction of the Wireless Network, EL shall perform such restoration, and the restoration of all work site areas to a clean and safe condition, upon the completion of each task, installation and phase of the work as set forth in the Construction Plans and schedules provided in Section 4, Network Construction; Construction Plans. Such

- restoration shall in each case be completed within a time reasonably required by the Streets Commissioner.
- 5.5.6 Removal of Communications Equipment. If EL removes the Communications Equipment from any Street Light(s) (except pursuant to Section 5.3, Removal for City Work, or Section 5.4, Emergency Action), then EL shall provide the Streets Commissioner written notice of such removal within thirty (30) days, specifying the Street Light(s) vacated, the exact locations thereof, and the date of removal. Removal of all Communications Equipment from any Street Light without its replacement or substitution by EL within thirty (30) days shall constitute a termination of EL's right to use such Street Light.
- 5.6 Standard of Performance; Risk. EL shall, at its own sole risk and expense, install and maintain Communications Equipment on Street Lights as authorized under this Agreement and Construction Permits issued by the City, and shall do so in a professional and workmanlike manner, in accordance with industry standards, and in accordance with all applicable federal, state, and local laws, ordinances and regulations. EL shall at all times maintain the Communications Equipment in a safe condition, and shall use and exercise due care, caution, skill and expertise in performing all work authorized under this Agreement and any Construction Permit issued by the City, and shall take all reasonable steps to safeguard work areas, including, without limitation, Street Lights, streets and sidewalks, City property and facilities, and property and facilities owned or occupied by others. The City shall exercise reasonable efforts, subject to the terms of this Agreement and the requirements of applicable provisions of The Philadelphia Code, to facilitate the issuance of appropriate permits and licenses to EL and/or its Subcontractor(s) for the installation and maintenance of the Communications Equipment pursuant to this Agreement.
- 5.7 <u>Electricity</u>. Nothing in this Agreement shall in any way obligate PAID or the City to pay the cost of electricity required by any Communications Equipment installed on Street Lights or in any other City site or facility.
- 5.8 Emergency Contact Information. EL and the City shall provide each other with current 24 hour, 7 days-a-week telephone and other contact information to facilitate prompt communications in emergencies. In the event that EL or the City reports an emergency to the other party, then the other party shall promptly respond and perform any work necessary to repair the emergency condition, to the extent required by its obligations hereunder, at no cost to the reporting party, unless the reporting party caused the emergency condition, or the reporting party is obligated to pay for such service by this Agreement or by applicable contracts between EL and the City.
- 5.9 <u>Information Tags</u>. EL will attach to each item of Communications Equipment installed on a Street Light an information tag that has the following information: Name and mailing address of the owner of the equipment; a telephone number that is staffed 24 hours a day, 7 days a week; the owner's facsimile number; and an email address where the owner's operating personnel can be reached.

- 5.10 <u>Hazardous Materials</u>. Unless the Streets Commissioner directs otherwise, EL shall provide written notice to the Commissioner of any material in its Communications Equipment which is defined or identified as hazardous, toxic, dangerous, or as creating a potential risk to human health or the environment, in any applicable federal, state, or local environmental law or regulation, before the material is brought onto any Street Light or any City property. For any such material, EL shall provide in its notice information sufficient to identify the material and enable the City to assess the hazard presented. EL shall undertake, at its cost, any and all investigatory, preventive, and remedial action (including emergency response, removal, containment and other remedial action) which may be required with respect to such material by any applicable federal, state, or local environmental laws or regulations, and/or by the orders of any governmental authority having jurisdiction under such laws or regulations. Compliance with any such order concerning the material which is directed to the City shall nonetheless be the responsibility of EL, at EL's cost.
- 5.11 Notice to Residents. If the Streets Commissioner reasonably determines that EL's installation of Communications Equipment on Street Lights in one or more blocks of the City will disrupt pedestrian or vehicular traffic, or adversely affect residents or businesses, then the Commissioner may require that EL provide notice to residents, and businesses in those blocks before commencing construction.
- 5.12 Compliance with Regulations. The Communications Equipment and all other work authorized and performed by EL under this Agreement shall be placed and maintained in accordance with the requirements and specifications of all applicable rules and regulations of the Federal Communications Commission, Pennsylvania Public Utilities Commission, the most current editions of the National Electrical Code, the National Electrical Safety Code, the rules and regulations of the Occupational Safety and Health Act (OSHA), and the rules and regulations of any governing authority having jurisdiction over the work to be performed. In the event such codes or authorities have differing requirements, the more stringent requirement(s) shall apply.
- 6 Interruption To Service And Damage To Property.
- 6.1 <u>Interruptions</u>. No Communications Equipment or installation thereof may in any way interfere with or degrade or interrupt the operation of any Street Light. If such interference, interruption, or degradation occurs, EL shall restore the Street Light and Street Light service within twenty-four (24) hours of notice by the Streets Commissioner. If EL fails to so correct, the Streets Department may take corrective action, upon notice to EL, and EL shall be liable for the reasonable costs incurred by the City therefor. Disruptions to Street Light operation caused by the attachment of electrical supply cables for Communications Equipment to the photocells on Street Lights, in accordance with the installation method set forth in Exhibit A, shall not be subject to this Section 6.1 provided the disruptions are necessary to install the Communications Equipment.
- 6.2 <u>Damage to Property</u>. EL, PAID and the City shall each assume responsibility for any loss from damage caused by that Party to the property of the other. Each party shall make an immediate report of the occurrence of any such damage to the other and, subject to Section 9.4, *Limitation of Liability*, shall on demand and receipt and review of documentation of

- the costs incurred, reimburse the other for its out of pocket cost incurred in making repairs or replacing damaged property to the extent the damages were caused by the negligence or wrongful actions of that party.
- 6.3 Nothing in this Section 6 shall (i) limit, alter, or waive EL's obligations under Section 9, *Indemnification, Independent; Contractor; Limitation of Liability*, or (ii) be construed as a waiver of those defenses, immunities, and limitations on damages available to the City and PAID pursuant to the Pennsylvania Political Subdivision Tort Claims Act, Act of October 5, 1980, P.L. 693, No. 142, (42 Pa.C.S.A. § 8541 et. seq.), or the Commonwealth Agency Law, 42 Pa. C.S.A. § 8521 et seq..
- 6.4 Parties Bear Their Own Costs. Except as expressly provided otherwise in this Agreement, EL and the City shall each pay the cost of the installation and maintenance of its own facilities. Nothing in this Agreement shall be construed to compel the City to retain in place or to maintain any Street Light for a period longer than the City requires for its street lighting purposes, provided, however, that if the City wishes to terminate its maintenance of a Street Light, the City shall cooperate reasonably with EL in obtaining access to alternative facilities in a manner that minimizes service interruptions and costs.

7 <u>Compensation</u>.

- 7.1 In consideration for the right to use Street Lights as provided in this Agreement, EL shall pay to PAID for the account of the City, during the Initial Term and any Renewal Terms, the money payments and provide to the City the other compensation set forth below. The money payments provided in Sections 7.1.2-7.1.3 below are the same money payments as those described in Section 1.1-1.5 of the Network Agreement.
- 7.1.1 <u>Prepayment of Street Light Use Fees</u>. EL shall pay to PAID for the account of the City the following non-refundable use fees for the Street Lights used by EL in operating the Wireless Network during the initial 10 year term of this Agreement:
- 7.1.1.1 Within twenty (20) days after the execution of this Agreement and satisfaction of the "First Payment Conditions," as provided in Section 4.1 of the Network Agreement, EL shall pay to PAID for the account of the City Two Hundred and Fifty Thousand Dollars (\$250,000).
- 7.1.1.2 Within twenty (20) days after Proof of Concept Acceptance, EarthLink shall pay to PAID for the account of the City Seven Hundred and Fifty Thousand Dollars (\$750,000).
- 7.1.1.3 One (1) year after Proof of Concept Acceptance, EarthLink shall pay to PAID for the account of the City One Million Dollars (\$1,000,000).
- 7.1.2 Monthly Street Light Use Fee.

In addition to the Street Light use fees provided in Section 7.1.1, EL shall pay to PAID for the account of the City a use fee of Two Dollars (\$2) per month for each Street Light

on which Communications Equipment of EL is installed as of the last day of the preceding month ("Monthly Use Fee"). Commencing with the month after Proof of Concept Acceptance, EL shall owe payments of the Monthly Use Fee in arrears for each previous month, and shall make such payments within sixty (60) days of the date that the City sends an invoice to EL for the fees. The City may, in the Streets Commissioner's discretion, invoice for the Monthly Use Fee monthly, quarterly, semi-annually, or annually.

7.1.3 City Wireless Accounts.

7.1.3.1 Wi-Fi Accounts.

EL shall furnish the number of Muni Wi-Fi product accounts to the City, at the City's sole option and for municipal purposes, for the prices provided in Sections 7.1.3.1 (a) – (c). The Muni Wi-Fi product accounts shall provide Base Service as set forth in Exhibits A and I of the Network Agreement. In the City's sole discretion, the City may package Core ISP Services, including but not limited to email, software, support, personal start pages, web hosting, newsgroups, VPN, and other Core ISP Services selected by the City with the Muni Wi-Fi product at the City's cost. The City will be responsible for the cost of CPE and installation services, and may, in its sole discretion, itself provide and/or charge its subscribers for CPE and installation services. Tier 0, 1, 2 and 3 subscriber support for the Muni Wi-Fi product accounts will be provided by the City or by WP, and Tier 3 network support will be provided by EL, all as set forth in Exhibit M of the Network Agreement. /EL shall provide the accounts and EL Tier 3 support directly to the City, and shall bill account charges, if any, and the cost of EL-provided support, if any, directly to the City. If EL discontinues the Muni Wi-Fi product and replaces it with a product of the same or higher transmission rate, as described in Exhibit I, Section 16 of the Network Agreement, EL shall provide accounts for such replacement product to the City in accordance with the terms of this Section 7.1.3. Any Muni Wi-Fi product accounts made available to the City will appear in EL's operational support systems as City wholesale accounts.

- (a) 1250 accounts will be furnished at no cost to the City.
- (b) The next 1750 accounts will be furnished at a price that does not exceed Four Dollars (\$4) per account per month, determined as follows: If the City and/or WP negotiate savings from PECO on electricity to power the Communications Equipment that results in the electricity cost falling below Four Hundred Thousand Dollars (\$400,000) per year, then the City will pay a price per account calculated according to the following formula:

Monthly Fee per Account = (0.5 x Annual Savings (i.e. the difference between \$400,000 and the lower price negotiated) / (12 x 1750)

(c) The City may purchase accounts in excess of Three Thousand (3,000) at the lowest wholesale rate charged to any wholesale customer on the network, without regard

to the number of accounts the City purchases (i.e. the City will receive the highest wholesale volume discount available to any wholesale customer on the network, regardless of number of accounts it purchases).

Accounts purchased under this subsection (c):

- (1) Shall be utilized for municipal government activities and purposes, and shall not be used for non-City activities, such as recreational use;
- (2) Shall fall under a wholesale account umbrella which shall be managed by a single designated point of contact within the City;
- (3) Shall access the system through City specific non-broadcast Service Set Identifiers (SSIDs);
- (4) Shall be routed by EL proxy RADIUS server to a designated City RADIUS server for authentication:
- (5) Shall be provisioned only to employees of the City or to the employees of City-Related Agencies as defined in Section 17-1401(9) of The Philadelphia Code;
- (6) Shall be accounts associated with the City's primary domain, phila.gov, or such other domains that as a matter of regular practice are managed by the City's IT departments;
- (7) Shall receive Tier 3 Network Operations Center (NOC) support via a single interface from the City's Tier 3 NOC to EarthLink's Tier 3 NOC support center;
- (8) Shall be billed on a single invoice each month;
- (9) Shall not be resold, except to City-Related Agencies as defined in Section 17-1401(9) of The Philadelphia Code;
- 7.1.3.2 T1 Alternative Accounts. EL shall furnish up to Seven Hundred (700) T1 alternative product ("Muni T1") accounts to the City, at the City's sole option and for the price of not more than One Hundred Dollars (\$100) per month per account, plus a one-time, non-recurring fee for CPE and installation (which may be waived by EL for a multiyear agreement commitment) of not more than Two Hundred and Fifty Dollars (\$250) per account. The Muni T1 alternative product shall provide point-to-multipoint fixed wireless data transmission and Internet access to municipal buildings and municipal government locations with a minimum 1.5 Mbps downstream / 1.5 Mbps upstream data transmission rate, and dynamic and static IP addresses (as requested by the City). EL shall exercise its best efforts to provide such data transmission rate(s). Muni T1 accounts shall be supported in the same manner as the Open Access ("OA") Wi-Fi Broadband Products, with Tier 0, 1, 2 and 3 subscriber support provided by the City or by WP, and Tier 3 network support provided by EL, all as set forth in Exhibit M of the Network Agreement. EL shall provide the accounts

- and deliver installation and CPE directly to the City, and shall bill account charges and the cost of EL provided CPE and installation services directly to the City.
- 7.1.3.3 Accounts at City Option. All Wi-Fi Broadband Product accounts and T1 Alternative Product accounts provided in this Section 7.1.3 will be furnished by EL to the City at the City's sole option, which option may be exercised by the City, in its sole discretion, at any time during the Initial Term of this Agreement and any Renewal Terms. EL shall make the accounts available to the City after Final Acceptance of the Wireless Network in accordance with the terms and conditions of this Section within thirty (30) days of the City's written request, or such lesser or greater time as EL and the City may agree to in writing. Neither PAID nor the City shall be obligated at any time to receive, use, or purchase any minimum number of wireless accounts.
- 7.1.3.4 If EL's provision of WiFi Retail Product accounts or T1 Alternative Product accounts under this Section 7.1.3 results in a sales tax charge to EL, the City will pay the amount of the sales tax to EL for remittance to appropriate tax authorities.
- 7.1.4 EL acknowledges that PAID has no economic interest in the payments due from EL under this Section 7.1.
- 7.2 Overdue Payments. If EL fails to pay any undisputed amounts payable under this Agreement within the time period provided for the payment in this Section 7, such unpaid amount shall bear interest thereafter until paid at the lower rate of eight percent (8%) per annum or, the maximum rate permitted by applicable law. This late payment interest has been agreed upon by PAID, the City and EL, after negotiation, as a reasonable estimate of the additional administrative costs and detriment that PAID and the City will incur as a result of any such failure by EL, the actual costs thereof being extremely difficult if not impossible to determine. The late payment interest constitutes liquidated damages to compensate PAID and the City for its damages resulting from such failure to pay and shall be paid to PAID, for the account of the City, together with such unpaid amount. Payment of such interest shall not excuse or cure any default by EL.
- 7.3 Audit. At all times during the Initial Term and any Renewal Term(s) of this Agreement, and for three (3) years after expiration or termination of the Agreement, EL shall keep and maintain all books, records, correspondence and documents of EL relating to the payment of the compensation provided in this Section 7, and on fourteen (14) days written notice shall make them available for audit, inspection and examination in Philadelphia (or another location if agreed upon by both parties in writing) by the Controller of the City of Philadelphia, PAID, and/or the Streets Commissioner or their designees. If such audit reveals a deficiency in payments due and owing under this Section 7, EL shall pay the deficiency within fourteen (14) days following written notice, together with interest thereon at a rate of ten percent (10%) per annum for each year in which the deficiency continued; provided, that the maximum amount of such interest shall be the greater of (i) ten percent (10%) per annum for each year the deficiency continued up to a maximum of four (4) years or (ii) ten per cent (10%) of the total amount of the deficiency, and provided further that such interest shall be in lieu of the interest required pursuant to subsection 7.2. If the amount of such deficiency is greater than ten percent (10%), then EL shall also pay

the costs incurred by the City and/or PAID for the audit. If any audit such reveals an overpayment by EL, EL shall be entitled to a credit against the next payments due and owing to the City under this Section 7 in the amount of the overpayment. Materials provided to the City under this Section 7.3 shall be subject to Section 13, *Confidentiality*.

7.4 Disposition of Payments on Dissolution or Bankruptcy of WP. In the event WP is dissolved, or if a Bankruptcy Event (as defined below) occurs with respect to WP, or WP otherwise ceases to operate, PAID shall have the right (but not the obligation), at its sole option, to receive, for the account of the City, all money payments and other consideration that EL is obligated to pay directly to WP under this Section 7 and under the Network Agreement, to the extent permitted by applicable law, or to designate an entity, including the City, to receive, for the account of the City, such money payments and other consideration. In the event PAID exercises the foregoing right, PAID will, to the extent permitted by law, assume the Network Agreement and/or the obligations of WP under the Network Agreement or cause its designee, if applicable, to assume the Network Agreement and WP's obligations under it. PAID further agrees that if it exercises such right, it or its designee, if applicable, will use the money payments and other consideration paid to it or its designee by EL to further the WP Mission. PAID may exercise the rights provided in this Section 7.4 by delivering written notice to EL, together with reasonable evidence that WP has been dissolved or has ceased to operate. If PAID designates another entity pursuant to this Section, PAID shall have no liability hereunder or with respect to the performance by such entity.

"Bankruptcy Event" means that WP (i) applies for or consents to the appointment of a receiver, trustee, liquidator or custodian or the like of itself or of its property, (ii) admits in writing its inability to pay its debts generally as they become due, (iii) makes a general assignment for the benefit of creditors, (iv) is adjudicated a bankrupt or insolvent, (v) commences a voluntary case under the United States Bankruptcy Code, or files a voluntary petition or answer seeking reorganization, an arrangement with creditors or an order for relief, or seeking to take advantage of any insolvency law or file an answer admitting the material allegations of a petition filed against it in any bankruptcy, reorganization, or insolvency proceeding, or action shall be taken by it for the purpose of effecting any of the foregoing, or (vi) has instituted against it, without its application, approval or consent, a proceeding in any court of competent jurisdiction, under any law relating to bankruptcy, insolvency, reorganization or relief of debtors, seeking in respect of WP an order for relief or an adjudication in bankruptcy, reorganization, dissolution, winding up, liquidation, a composition or arrangement with creditors, a readjustment of debts, the appointment of a trustee, receiver, liquidator or custodian or the like of WP or of all or any substantial part of its assets, or other like relief in respect thereof under any bankruptcy or insolvency law, and, if such proceeding is being contested by WP in good faith, the same shall (A) result in the entry of an order for relief or any such adjudication or appointment or (B) remain unvacated, undismissed and undischarged for a period of 90 days.

7.5 The payments provided in Section 7.1 shall be deemed to include payment of fees related to the issuance of Construction Permits (but not to include the costs of construction inspections provided in Section 5.1.1.2); provided, that the Streets Commissioner may

charge a permit fee of up to Eighteen Dollars (\$18) per Street Light for Construction Permits issued to EL for any Street Light in excess of four thousand (4,000).

- 8 Insurance; Removal Security.
- 8.1 Insurance.
- 8.1.1 <u>General Requirements</u>.

EL shall, at its sole cost and expense, procure and maintain in full force and effect, covering risks associated with the construction, installation, placement, operation, maintenance, use, renewal, replacement, relocation, repair and removal of Communications Equipment placed or to be placed on Street Lights or otherwise on City property or in the Right-of-Way, the types and minimum limits of liability insurance specified below. All insurance shall be procured from insurers (i) licensed and authorized to write insurance in Pennsylvania, (ii) acceptable to the City's Risk Manager and to PAID, which acceptance will not be unreasonably withheld or delayed, and (iii) with an A.M. Best rating of A-VIII or higher, unless agreed otherwise in writing by the City's Risk Manager and PAID. All insurance required herein shall be written on an "occurrence" basis, or if any policy is written on a claims made basis, the retroactive date may not be advanced beyond the date of this agreement and coverage shall be maintained in full force and effect for two (2) years after termination of this Agreement, which coverage may be in the form of tail coverage or extended reporting period coverage if agreed by the parties. The certificate of insurance shall provide for at least thirty (30) days prior written notice to be given to the City Department of Streets (7th Floor, Municipal Services Building, Philadelphia, PA 19102), the City's Risk Manager (One Parkway, 14th Floor, 1515 Arch Street, Philadelphia, PA 19102-1595), the City's Law Department (One Parkway, 17th Floor, 1515 Arch Street, Philadelphia, PA 19102-1595); and to PAID and the Philadelphia Industrial Development Corporation ("PIDC") (Philadelphia Authority for Industrial Development, 2600 Centre Square West, 1500 Market Street, Philadelphia, PA 19102-2126, Attn: Risk Magement – Wireless Phl) in the event coverage is cancelled or non-renewed. In the event coverage is materially changed, EL or its insurer shall provide at least thirty (30) days prior written notice to the City and to PAID at the above addresses.

- 8.1.2 The City of Philadelphia, PAID and PIDC, and their respective officers, directors, and employees, shall be named as endorsed named insureds on each insurance policy required by this Agreement, except policies for Workers' Compensation, and on any insurance EL requires of any person, agent, Subcontractor, or employee engaged or employed in, about or upon the work by, at the instance of, or with the approval or consent of EL, as respects the acts, omissions, operations and activities of, or on behalf of, the named insured, in regard to products supplied, or work or services performed for, or under an agreement with, PAID and/or the City of Philadelphia.
- 8.1.3 A certificate of insurance shall be provided to PAID and to the City's Risk Manager, as set forth below, that evidences the required coverages. The certificate of insurance shall

- state that such coverages will be primary to any coverage available to them and will be noncontributory.
- 8.2 <u>Types and Minimum Limits of Insurance</u>. EL shall furnish the following types and minimum limits of insurance:
- 8.2.1 Workers' Compensation and Employers' Liability insurance
 - (1) Workers' Compensation: To meet Pennsylvania statutory requirements
 - (2) Employers' Liability: \$500,000 each Accident Bodily Injury by Accident; \$500,000 Each Employee Bodily Injury by Disease; and \$500,000 Policy Limit Bodily Injury by Disease
- 8.2.2 Commercial General Liability insurance
 - (1) Limit of liability: The Commercial General Liability Insurance Policy in an amount not less than \$5,000,000 per occurrence, Twenty Million Dollars (\$20,000,000) annual aggregate, written on an occurrence form, provided this coverage may be maintained by EL combining primary and excess policies to meet these requirements.
 - (2) Types of insurance: Applicable dollar limit, as set forth in (1), per occurrence combined single limit for bodily injury (including death) and property damage liability; applicable dollar limit, as set forth in (1), for personal and advertising injury; applicable dollar limit, as set forth in (1), general aggregate; and applicable dollar limit, as set forth in (1), for products and completed operations
 - (3) Coverage: Premises operations; blanket contractual liability; personal injury liability (employee exclusion deleted); products and completed operations; independent contractors; employees and volunteers as additional insureds; cross liability; broad form property damage (including completed operations); and explosion, collapse, underground hazards (XCU)
- 8.2.3 Commercial Automobile liability insurance
 - (1) Limit of liability: \$5,000,000 per occurrence combined single limit for bodily injury (including death) and property damage liability, provided this coverage may be maintained by EL combining primary and excess policies to meet these requirements
 - (2) Coverage: Owned, non-owned and hired vehicles
- 8.2.4 The City's Risk Manager, with the consent of PAID, is hereby authorized to reduce the requirements set forth herein in the event he determines that such reduction is in Philadelphia's best interest.

8.3 Evidence of Insurance Coverage

Prior to commencing any work under an agreement with WP or accessing or using any Street Light, EL shall deliver to PAID and the City of Philadelphia certificates of insurance evidencing the required coverages and endorsed named insureds, which must specifically reference in the description sections of the certificates such agreement between WP and EL and this Street Light Use Agreement. Original certificates of insurance must be submitted to PAID at the following address:

Philadelphia Authority for Industrial Development 2600 Centre Square West 1500 Market Street Philadelphia, PA 19102-2126 Attn: Risk Management – Wireless Phl

and to the City's Risk Manager at the following address:

City of Philadelphia Finance Department Division of Risk Management 1515 Arch Street, 14th Floor Philadelphia, PA 19102-1579 (Fax No.: 215-686-1705).

PAID and the City each reserves the right to inspect on the site of EL copies of policies of all insurance required in this Section 8, or alternatively, at the option of the City or PAID, to require EL to demand that its authorized insurance representatives furnish written responses to all inquiries made by the City or PAID pertaining to the insurance required under the Agreement at any time upon ten (10) days written notice to such insurance representatives. Also, in the event that (i) a claim is filed against PAID, PIDC, or the City, or their officers, employees, agents, or representatives, for which PAID, PIDC, or the City believes, in good faith, there may be coverage under an insurance policy required by this Agreement, and (ii) PAID, PIDC, or the City has tendered such claim to the insurer that issued such policy, and (iii) such insurer has not agreed, within ninety (90) days, to cover such claim, including defense and indemnity, without reservation, then the City or PAID or PIDC each has the right to demand of EL, and to receive within thirty (30) days, a copy of such insurance policy.

8.4 Contractors.

EL shall require that its contractors procure and maintain, or cause to be procured and maintained, throughout the entire period of construction, operation and maintenance of Communications Equipment on any Street Light, the types of insurance, limits of insurance, and designation of additional insureds which are required of EL under this Section 8.

- 8.5 Removal Security. Prior to Proof of Concept Acceptance, EL shall furnish to the City and PAID removal security in the form of a letter of credit, in the amount of Five Hundred Thousand Dollars (\$500,000), to pay the City's costs of removing the Communications Equipment from Street Lights in the event EL fails to remove the Communications Equipment when required to do so under this Agreement. EL shall deliver to the City and PAID a letter of credit issued by Bank of America or another national bank approved by the City and PAID, in the form attached hereto as Exhibit C, with changes as approved by the City and PAID.
- 8.5.1 The letter of credit, or replacement letters of credit delivered to the City at least ten (10) days before the expiration of the previous letter of credit, shall be effective throughout the Initial Term and any Renewal Terms and for two years after the date of expiration or termination of the last such Term; provided, that the letter of credit shall be returned to EL for cancellation prior to such time if, but only if, (i) the City determines in its reasonable judgment that all Communications Equipment has been removed from Street Lights, and (ii) the City determines in its reasonable judgment that all storage obligations of the City regarding removed Communications Equipment that are provided in this Section 8.5.1 and elsewhere in the Agreement have also terminated.
- 8.5.2 The City shall be entitled to draw on the letter of credit five (5) business days following delivery of a written certification by the Streets Commissioner of the City to the issuing bank and to EL certifying that EL has not met its obligation to remove Communications Equipment from the Street Lights pursuant to this Agreement within the time required under the Agreement, and certifying the dollar amount required by the City to remove the Communications Equipment, together with reasonable documentation. The letter of credit disbursement shall be limited to the amount of the removal expenses stated and reasonably documented by the Streets Department. After a draw request is paid by the issuer of the letter of credit in that amount, then the balance remaining on the letter of credit shall be available for further draws by the City as may be required to remove other Communications Equipment not removed by EL pursuant to the Agreement. In the event the City's actual cost of removal is less than the amount stated in the Streets Commissioner's certification, or in the event the City does not remove the Communications Equipment within eighteen (18) months of the disbursement, then the City shall promptly refund the amount of such difference to the issuer of the letter of credit, which will credit the refund to the amount that can be drawn on the letter of credit.
- 8.5.3 Except as otherwise expressly stated in the letter of credit, the credit provided therein is subject to the International Standby Practices (ISP98), International Chamber of Commerce Publication No. 590 and, to the extent not in consistent therewith, the laws of the State of New York.
- 9 Indemnification; Independent Contractor; Limitation of Liability.
- 9.1 <u>Independent Contractor</u>. EL and its Subcontractors are each an independent contractor and shall not in any way or for any purpose be deemed or intended to be an employee or agent of PAID or the City. Neither EL nor its Subcontractor(s) nor their respective employees,

- directors, agents or Subcontractors shall in any way represent that they are acting as employees, officials or agents of PAID or the City.
- 9.2 <u>Indemnification</u>. EL shall indemnify, defend and hold harmless the City, PAID and PIDC and their respective officers, employees, contractors, consultants, members and agents, from and against any and all losses, costs (including, but not limited to, litigation and settlement costs and counsel fees), claims, suits, actions, damages, liabilities and expenses, to the extent caused by or resulting from EL's act or omission or negligence or fault, or the act or omission or negligence or fault of EL's agents, Subcontractors, suppliers, or employees in connection with this Agreement, use of the Street Lights by EL or its agents, Subcontractors, suppliers, or employees, and the installation, repair, replacement, operation, maintenance, and/or use of Communications Equipment on Street Lights, including, but not limited to, those in connection with loss of life, bodily injury, personal injury, damage to property, contamination or adverse effects on the environment, intentional acts, failure to pay such Subcontractors or suppliers.
- 9.3 <u>Indemnification for Infringement of Proprietary Rights.</u>
- 9.3.1 EL will indemnify PAID, PIDC, the City, and their respective officers, employees, contractors, consultants, members and agents, but not any customers or third parties ("Indemnitees"), from losses, costs and damages and defend any suit or proceeding brought against the Indemnitees to the extent that the suit or proceeding is based on a claim that the Network Operations constitute infringement of any United States patent, mask work, trade secret or copyright and EL will pay, or EL will cause the supplier of the infringing Network Operations to pay, all damages awarded by final judgment (from which no appeal may be taken) against the Indemnitees, but not any customers or third parties, holding that the Network Operations do so infringe, on condition that EL (i) is promptly informed and furnished a copy of each communication, notice or other action relating to the alleged infringement, (ii) is given authority, information and assistance necessary to defend or settle the suit or proceeding in a manner as EL may determine and (iii) is given sole control of the defense (including the right to select counsel and to require and allow the supplier of the infringing Network Operations to pay damages and defend as provided herein), and the sole right to compromise and settle the suit or proceeding; provided, however, that no such compromise or settlement shall impose any liability or burden on the City or PAID without their consent. If any Network Operations are held to infringe any valid United States patent, mask work, trade secret or copyright as set forth above, and an injunction from a court of competent jurisdiction permanently prohibits the use of the same in the Wireless Network, or if EL believes infringement is likely, then EL will (a) procure the right to use the Network Operations in the Wireless Network free of any liability for infringement, (b) replace or modify the Network Operations in the Wireless Network with a non-infringing substitute otherwise complying substantially with the specifications for the Wireless Network; or (c) comply with the subsequent sentence. In the event that (i) a court of competent jurisdiction determines that the Network Operations infringe the above-defined rights of another party, (ii) all right of appeal or stay is exhausted as to such order, and (iii) EL is unable to replace or modify such infringing Network Operations in the Wireless Network or otherwise procure the right to continue using the Network Operations in the Wireless Network then

EL will be entitled to remove the infringing Network Operations from the Wireless Network and obtain substitutes that will result in the Wireless Network performing substantially as required in this Agreement. "Network Operations" means (i) the Wireless Network or any portion thereof (ii) products and services provided directly by EL, parents and subsidiaries of EL and companies commonly owned with EL ("Affiliates") or Subcontractors through the Wireless Network, and (iii) operation and design (including attaching devices to Street Lights) of the Wireless Network.

- 9.3.2 <u>Limitation of Intellectual Property Indemnification</u>. In no event will EL or its suppliers be liable for an infringement that: (i) would not have existed but for the use of products or services in the Wireless Network that were not provided by EL or its Affiliates or Subcontractors; or (ii) would not have existed but for modification of, or additions to, the Network Operations by the City or its contractors or agents. Prior to a final determination that the infringement arose from clauses (i) or (ii) in the preceding sentence, EL shall remain fully responsible, at its expense, for the defense and indemnification of any infringement claim in accordance with this Section: provided, that if it is finally determined that the infringement arose from clauses (i) or (ii) in the preceding sentences, then PAID and/or the City will reimburse EL for its reasonable expenses (including reasonable litigation costs and attorneys fees) incurred therein.
- 9.3.3 Exclusive Liability for Infringement. THE FOREGOING CONTAINS THE SOLE AND EXCLUSIVE LIABILITY FOR EL'S AND ITS SUPPLIERS' INDEMNITY AGAINST INFRINGEMENT IN THIS SECTION 9.3, WHETHER DIRECT OR CONTRIBUTORY, AND IS IN LIEU OF ALL WARRANTIES, EXPRESS, IMPLIED OR STATUTORY IN REGARD THERETO, INCLUDING, WITHOUT LIMITATION, THE WARRANTY AGAINST INFRINGEMENT SPECIFIED IN THE UNIFORM COMMERCIAL CODE.

9.4 <u>Limitation of Liability</u>.

- 9.4.1 Subject to the exceptions provided in Section 9.4.2, EL's liability to the City and/or PAID for damages in connection with this Agreement shall not exceed the total amount of One Million Dollars (\$1,000,000), and the liability of the City and PAID to EL for damages in connection with this Agreement shall not exceed the total amount of One Million Dollars (\$1,000,000).
- 9.4.2 Section 9.4.1 and the limitations of liability provided therein shall not apply to damages relating to or arising from loss of life, bodily injury, personal injury, damage to property, contamination or adverse effects on the environment, or EL's liability for damages pursuant to Section 9.3, *Indemnification for Infringement of Proprietary Rights*.
- 9.4.3 Notwithstanding any provision or obligation to the contrary set forth in this Agreement or in any related document, no provision herein shall be construed so as to give rise to any pecuniary liability of PAID or its members, directors, officers, employees, agents, subcontractors, or representatives for a charge upon or claim against the general credit or assets of PAID or of such members, directors, officers, employees, agents, subcontractors, or representatives except to the extent of PAID's economic interest in this Agreement as set forth in the next sentence. Any pecuniary liability of PAID in connection

- herewith, under any theory of law, including contract or tort, whether for damages for breach or for any other reason whatsoever, is expressly and exclusively limited to PAID's economic interest in this Agreement and the lien of any judgment shall be expressly restricted thereto.
- 9.5 <u>Statutory Immunities and Defenses</u>. Nothing in this Section 9 shall waive or amend any defense or immunity that the City or PAID or their officers, agents or employees may have under the Pennsylvania Political Subdivision Tort Claims Act, 42 Pa.C.S.A. section 8541, et seq., or the Commonwealth Agency Law, 42 Pa. C.S.A. § 8521 et seq.
- 9.6 No damages for Delay. EL shall be entitled to a reasonable extension of the time for performing any action required of EL under this Agreement if its failure to perform such action is caused by City delay; provided, however, that no such extension shall be granted unless a written request, setting forth in reasonable detail the grounds therefor, is received by the City within ten (10) calendar days following the events giving rise to such alleged delay, or at least two (2) days prior to the expiration of the time for performance if it is less than ten (10) days. EL's sole and exclusive remedy for costs or damages arising from any delay to its performance of actions required under this Agreement or under the Network Agreement that is caused by any act or omission of the City (including its officers, employees, contractors and agents) shall be an extension of the time of performance, as provided in this Section 9.6 (and in the Network Agreement for actions required under the Network Agreement). In no event shall EL be entitled to monetary payment or compensation of any kind from PAID or the City for any costs or damages incurred as a result of delay in its performance of any obligation of EL, PAID or the City, or to WP, EL's contractors, or any other party, that is caused by any act or omission of the City or PAID (including their officers, employees, contractors, members and agents), including, without limitation, such delay to the construction of the Wireless Network, and/or delay in the performance of any other obligation of EL under the Network Agreement or under this Agreement.

10 Dispute Escalation.

- 10.1 Any dispute between EL and the City or PAID which arises during the Initial Term or any Renewal Term of this Agreement and which the parties to the dispute cannot then resolve, shall be subject to the following dispute resolution procedure.
- 10.2 The parties shall attempt to resolve any controversy claim, problem or dispute (a "Dispute") arising out of, or related to, this Agreement through good faith consultation in the ordinary course of business. In the event that the Dispute is not resolved by the project managers of the parties to the Dispute, any party to the Dispute may upon written notice to the other party or parties to the Dispute request that the matter be referred to senior management officials within each respective organization with express authority to resolve the Dispute ("Request for Internal Resolution"). A written Request for Internal Resolution shall be given by any party to the Dispute within thirty (30) calendar days of the party's knowledge of the unresolved Dispute. Senior management officials shall meet or confer at least once in good faith, to negotiate a mutually acceptable resolution within thirty (30) days of the Request for Internal Resolution ("Dispute Resolution Period"). For the purposes of this paragraph, the designated project manager or senior management official

for PAID shall be Samuel V. Rhoads, Senior Vice President of PIDC; for the City, shall be the Managing Director of the City or his designee; and for EL, shall be the President of EL Municipal Networks. The project managers and senior management may be changed by notice given by the party changing its personnel. EL acknowledges that any disputes are likely to be between it and the City and that PAID is not intended to be a necessary participant in any such dispute.

- 10.3 Project managers and senior management officials are required to meet only once but may mutually agree to meet more than once if it appears that further meetings may successfully resolve the Dispute.
- 10.4 The provisions of this Section 10 shall not apply to a Termination Default as set forth in Section 12, *Termination and Default*, though the executives of each party shall make themselves available to each other upon request during the cure periods set forth therein.
- 11 <u>Term, Renewal, Expiration and Removal.</u>
- 11.1 <u>Initial Term</u>. This Agreement shall become effective on the Effective Date and unless sooner terminated in accordance with the terms and conditions hereof, shall continue in effect for ten (10) years after the date on which the Proof of Concept Trial is accepted and approved by the City and WP (the "Initial Term").
- 11.2 <u>Renewal</u>. Upon expiration of the Initial Term, the Agreement may be renewed for up to two (2) successive five (5) year periods of time ("Renewal Term(s)") at the discretion of PAID, at the direction of the City, and with the agreement of EL, provided that EL furnishes to PAID and the City, for each Renewal Term, at least eighteen (18) months prior written notice of its intent to renew. If PAID elects not to renew the Agreement, it will so notify EL at least twelve (12) months prior to the expiration of the then current term.
- 11.3 Expiration and Removal. Upon expiration of the Initial Term and all Renewal Terms entered into pursuant to Section 11.2, EL shall, at its own expense, within twelve (12) months of such expiration, remove the Communications Equipment from the Street Lights and restore all Street Lights to at least the condition (taking into account normal wear and tear) they were in prior to the installation of the Communications Equipment. If EL fails to remove the Communications Equipment from the Street Light Poles within twelve (12) months, the City shall have the right, but not the obligation, to remove any of the Communications Equipment so remaining, and EL shall reimburse the City for its costs incurred in connection with removing the Communications Equipment; provided, that if EL does not so reimburse the City within thirty (30) days following the City's written notice of its removal costs, the City may draw on the removal security provided in Section 8.5, Removal Security; and provided further that if sufficient funds remain of such removal security to pay storage costs after the payment of all costs incurred by the City to remove the Communications Equipment, the City will store the removed Communications Equipment in a location of the City's choice and give EL written notice that it has done so. If EL does not remove the Communications Equipment from the City's storage location within sixty (60) days following such notice, the City may dispose of the stored

Communications Equipment in any manner that it chooses, without cost or liability to the City. In no event shall the City be liable for theft or loss of or damage to the Communications Equipment, of any kind or nature whatsoever, that occurs during removal from the Street Lights, or while the Communications Equipment is in storage, whether or not such theft, loss, or damage is the result of any act or omission of the City.

12 Default and Termination.

- 12.1 Termination for Termination Default. PAID, at the direction of the City, shall have the right, but shall not be obligated to, terminate this Agreement and EL's authorization to use Street Lights provided in the Agreement in the event that one or more of the defaults set forth in Sections 12.1.1-12.1.3 ("Termination Default(s)") has occurred and continued uncured until after the end of all applicable cure periods set forth therein. Notwithstanding anything to the contrary in this Agreement, PAID shall not, however, terminate the Agreement and EL's authorization to use Street Lights provided in the Agreement unless one or more of the Termination Defaults has occurred and continued uncured until after the end of all applicable cure periods set forth therein; and any other default or failure of EL to comply with this Agreement shall not permit such termination. In the event of conflict with any other provision of this Agreement relating to termination of the Agreement and such authorization to use Street Lights, this provision shall govern; provided, however, that nothing in this Section 12.1 is intended to or shall limit, alter, or waive any remedy provided in the Agreement other than termination of the Agreement and EL's authorization to use Street lights, including, without limitation, the termination and other remedies set forth in Section 12.2, Remedies for Noncompliance, and the remedies provided elsewhere in the Agreement.
- 12.1.1 Failure to Accomplish System Implementation and Deployment. It shall be a Termination Default by EL if deployment of the Wireless Network in substantially all of the Coverage Area (as described in the Design Control Documents) is not accomplished within twelve (12) months after the date set forth on the Schedule for completion of the implementation and deployment of the network that is provided for in the Network Agreement; provided, that such time will be extended by any delays caused by WP, PAID, the City, by providers of electricity to the Wireless Network, providers of telecommunications service to the Network other than EL, by third parties (other than Subcontractors of EL) that are not under the control of EL, or by a Force Majeure Event. Provided that Communications Equipment to be installed on Street Lights, and other System elements specified in the Design Control Documents, have been deployed and activated in substantially all of such Coverage Area, the parties agree that Wireless Network performance problems and configuration problems and the inability of computers to receive wireless internet access from the Network will not constitute a Termination Default under this Section 12.1.1.
- 12.1.2 <u>Failure to Pay Street Light Use Fees</u>. It shall be a Termination Default by EL if EL fails to pay any of the payments provided in Section 7.1.1, *Prepayment of Street Light Use Fees* or in Section 7.1.2, *Monthly Street Light Use Fee*, within sixty (60) days after written notice of failure to pay when due is delivered to EL, provided that the notice must

state that the failure to pay, if not cured, will constitute a Termination Default under this Agreement.

- 12.1.3 Network Outage. The following shall constitute a Termination Default by EL: If less than twenty percent (20%) of the receiving/transmitting nodes in the Wireless Network are capable of connecting Subscribers to the Internet for an entire calendar day, the Network will be considered "Dark" and that day will be considered a "Dark Day"; provided, that such day will not be considered a Dark Day if the failure to satisfy the above test was caused by WP, PAID, the City, providers of telecommunications service to the Network other than EL, providers of electricity to the Network, or by third parties (other than Subcontractors of EL) not under the control of EL, or by Force Majeure Events. If the System is continuously Dark for a period of thirty (30) or more days, PAID, at the direction of the City, may send a notice of termination, which shall be effective ninety (90) days after receipt by EL if such default has not been cured by the Network operating without any Dark Days for thirty (30) days continuously. If EL does cure the default by the Network operating without any Dark Days for thirty (30) days continuously prior to the end of such ninety (90) day period, then the termination shall not be effective and the defaults shall be cured. If EL is unable to cure the default within said ninety (90) days by the Network operating without any Dark Days for thirty (30) days continuously, then this Agreement and EL's authorization to use the Street Lights under the Agreement shall terminate twelve (12) months after the end of said ninety (90) day cure period; provided, that during such twelve (12) months EL will be entitled to continue operating the Network and the Communications Equipment on the Street Lights will not be removed, except as provided elsewhere in this Agreement. In addition, during such twelve (12) months, EL shall be entitled to sell the Communications Equipment; and provided the City, PAID and WP approve the purchaser and the assignment of EL's rights and interest in the Network (as provided in Section 17.16, Assignment, of this Agreement and in the assignment provision of the Network Agreement) to the transferee, EL shall also be entitled to transfer to a third party (subject to the terms of the foregoing assignment provisions) the Network, its ongoing operations, and EL's rights under this Agreement.
- 12.1.4 Removal of Communications Equipment Upon Termination for a Termination Default. Upon termination of this Agreement for a Termination Default, EL shall, at its own expense, within twelve (12) months of the date of termination, remove the Communications Equipment from the Street Lights and restore all Street Lights to at least the condition (taking into account normal wear and tear) they were in prior to the installation of the Communications Equipment. If EL fails to remove the Communications Equipment from the Street Light poles within twelve (12) months, the City shall have the right, but not the obligation, to remove any of the Communications Equipment so remaining, and EL shall reimburse the City for its costs incurred in connection with removing the Communications Equipment; provided, that if EL does not so reimburse the City within thirty (30) days following the City's written notice of its removal costs, the City may draw on the removal security provided in Section 8.5, Removal Security; and provided further that if sufficient funds remain of such removal security to pay storage costs after the payment of all costs incurred by the City to remove the Communications Equipment, the City will store the removed Communications

Equipment in a location of the City's choice and give EL written notice that it has done so. If EL does not remove the Communications Equipment from the City's storage location within sixty (60) days following such notice, the City may dispose of the stored Communications Equipment in any manner that it chooses, without cost or liability to the City. In no event shall the City be liable for theft or loss of or damage to the Communications Equipment, of any kind or nature whatsoever, that occurs during removal from the Street Lights, or while the Communications Equipment is in storage, whether or not such theft, loss, or damage is the result of any act or omission of the City.

12.2 Remedies for Noncompliance. In the event EL's use of a Street Light (including, without limitation, its installation, repair, replacement, operation, or maintenance of Communications Equipment on the Street Light) does not comply with the terms and conditions of the Sections listed below in this Section 12.2, PAID, at the direction of the Streets Commissioner, shall have the right to exercise any one or more of the remedies provided in Sections 12.2.1-.3, provided that EL has not corrected the noncompliance within fourteen (14) days following written notice from the Commissioner to EL describing the noncompliance in reasonable detail (or such longer period as the Commissioner may approve). The Commissioner shall have the right to exercise such remedies notwithstanding anything to the contrary in this Section 12 or elsewhere in the Agreement, and in addition to, not in lieu of, any other remedies available to the City and/or PAID at law or in equity (including, without limitation, application to a court of competent jurisdiction sitting in Philadelphia County for an injunction to enforce compliance), and in addition to, not in lieu of, any other rights of the City and/or PAID provided in this Agreement.

The Sections of the Agreement to which this Section 12.2 applies are:

- Section 2.4, Authorized Communications Equipment
- Section 2.5, Conditions of City Issuance of Construction Permits
- Section 2.6, Maximum Number of Street Lights Authorized
- Section 3, Application and Approval to Place Communications Equipment
- Section 4, Construction Plans; Organization of the Work
- Section 5, *Installation and Maintenance*
- Section 6, *Interruption to Service and Damage to Property*
- Section 8, *Insurance*
- 12.2.1 <u>Termination</u>. The Streets Commissioner may terminate EL's right to use a Street Light if EL's use of the Street Light, does not comply with the terms and conditions of the Sections of this Agreement set forth above as Sections to which Section 12.2 applies. EL shall remove its Communications Equipment from the Street Light within ten (10) days after receipt of the Commissioner's written notice of termination; if it does not do so, the Streets Department may remove the Communications Equipment at EL's expense and without liability to EL for loss or damage to the Communications Equipment. EL may not again use the Street Light without first obtaining a Construction Permit therefor, but nothing in this Section 12.2.1 will prohibit EL from applying for and obtaining a Construction Permit for a Street Light as to which it has been terminated.

- 12.2.2 <u>Denial of Construction Permits</u>. The Streets Commissioner may deny Construction Permits applied for after the expiration of the cure period(s) provided herein, provided the non-compliance(s) for which this remedy is exercised are persistent and uncured after the applicable cure periods.
- 12.2.3 <u>Money Payments</u>. The Streets Commissioner may require EL to pay a sum of not more than seven hundred dollars (\$700) for each event of noncompliance that is not cured after the expiration of the cure period provided herein. A separate and distinct event of noncompliance, subject to the foregoing payment, shall be deemed committed on each day on which noncompliance occurs or continues.
- 12.3 <u>Non-Termination Defaults</u>. The following shall be events of default under this Agreement for which a non-defaulting party may exercise the remedies provided in Section 12.4, but only those remedies ("Non-Termination Defaults"). Termination of this Agreement and (in the case of EL) EL's authorization to use the Street Lights provided in the Agreement shall not be a remedy for any Non-Termination Default.
- 12.3.1 <u>Failure to Meet Network Service Levels</u>. The "non-termination defaults" provided in Section 16.4 of the Network Agreement, relating to Network service levels, shall also be Non-Termination Defaults under this Agreement.
- 12.3.2 EL's material failure to comply with any provision, term, or condition of this Agreement other than the terms set forth in Section 12.3.1, *Failure to Meet Network Service Levels*, or in the Sections of the Agreement identified in Section 12.1, *Termination for Termination Default* or Section 12.2, *Remedies for Noncompliance*, for which the default remedies are provided herein; provided, that such failure to comply shall not be an event of Non-Termination Default unless it remains uncured for thirty (30) days following delivery of the City's written notice of default.
- 12.3.3 PAID's material failure to comply with any provision, term or condition of this Agreement; provided, that such failure to comply shall not be an event of Non-Termination Default unless it remains uncured for thirty (30) days following delivery of EL's written notice of default.
- 12.4 Remedies for Non-Termination Default.
- 12.4.1 If an event of Non-Termination Default occurs, the non-defaulting party shall be entitled to exercise every remedy, other than termination of this Agreement (and in the case of EL other than termination of the authorizations to use the Street Lights provided in this Agreement), available to it under the law and under the Agreement, including without limitation specific performance, injunctive relief, and money damages; provided, however, that EL shall not have the right to exercise the remedies of specific performance or injunctive relief for any breach by PAID or the City of Section 2.2, *Authorization Not Exclusive*.
- 12.4.2 <u>Equitable Relief</u>. Each party agrees that any threatened or actual Non-Termination Default may cause immediate irreparable harm to the non-defaulting party for which

there is no adequate remedy at law. Accordingly, each party agrees that the nondefaulting party will be entitled to an order for specific performance and to injunctive relief from a court of competent jurisdiction as remedy for any threatened or actual Non-Termination Default. Except for the limitations on termination and injunctive relief that are stated in this Section 12, nothing herein will limit either party's right to any other remedies at law, including the recovery of money damages from the defaulting party. Nothing in this Section 12 or in Section 9.4, *Limitation of Liability*, will prevent or prohibit either party from obtaining an order of specific performance and an injunction from a court of competent jurisdiction requiring the defaulting party to comply with the provisions of the Agreement, and not to take actions that violate its obligations under the Agreement; provided, that such order shall not prevent EL from operating the Wireless Network and maintaining the Communications Equipment on the Street Lights, except in the event of termination for Termination Default or on account of expiration of the Initial Term and any Renewal Terms; and provided further that such order for specific performance and injunction shall not require either party to pay to the other party any sums in excess of the liability limitation set forth in Section 9.4. Notwithstanding anything to the contrary in this Section 12.4.2 or in Section 12.4.1, the Parties agree that EL shall not have the right to exercise the remedies of specific performance or injunctive relief for any breach by PAID or the City of Section 2.5, Authorization Not Exclusive.

12.4.3 <u>Termination by EL</u>. EL shall not terminate this Agreement or the Network Agreement, for default or otherwise, except that if the Network Agreement terminates pursuant to Sections 3.2(d), 4.4, 4.5, 10.3, 12.1, 16.3, 16.6 or 16.7 thereof, then this Agreement shall also terminate.

13 <u>Confidentiality</u>.

- 13.1 Except as permitted by this Agreement, neither Party will use the Confidential Information of the other. Each Party will use reasonable efforts, to the extent it does for its own proprietary and confidential information of like nature, to prevent any Confidential Information of the other Party from being disclosed to third parties except as set forth in this Agreement or as authorized by a Party in writing. EL acknowledges that as of the date of execution of this Agreement, it has not provided any Confidential Information to PAID and does not intend to provide any such information in the future. EL further acknowledges that PAID is a governmental agency subject to sunshine and Right-to-Know Act laws; PAID's obligations under this Section 13 are limited in all respects by applicable laws.
- 13.2 "Confidential Information" will mean a Party's trades secrets under applicable law and also the proprietary and confidential information disclosed to the other Party. PAID acknowledges that all information concerning EL, its business plans, pricing, proprietary rights, Subscribers, customers and suppliers will be deemed Confidential Information of EL, whether or not marked "CONFIDENTIAL", and will not be used or disclosed by PAID except in the proper performance of its obligations imposed by law or under this Agreement. Subject to applicable law, any obligation contained herein that requires the disclosure of information of a third Party shall not be required if it would violate an applicable agreement of a Party with that third Party. Each Party agrees that this Section is

intended to protect the Confidential Information of the providers and Subcontractors of EL.

- 13.3 This Section 13 will not apply to any information which: (A) is or becomes public knowledge through no wrongful act of the receiving Party; (B) is already known to the receiving Party without obligation of confidentiality; (C) is rightfully obtained by the receiving Party from any third Party without similar restriction and without breach of any obligation owed to the disclosing Party; (D) is independently developed by the receiving Party; (E) is furnished to a third Party by the disclosing Party without a similar restriction on the third Party's rights; (F) is disclosed pursuant to a lawful requirement or request of a governmental agency; (G) is approved for release by written authorization of the disclosing Party; or (H) is disclosed in an enforcement or defense in a court of law pursuant to this Agreement or as is appropriate to assert and protect the rights of either Party. Despite the restrictions of this Section 13, EL and PAID shall each be entitled to provide the other's Confidential Information to third Party contractors and consultants that have executed a written confidentiality agreement containing substantially the provisions of this Section 13.
- 13.4 The provisions of this Section 13 shall survive for any trade secret for so long as such information qualifies as a trade secret under applicable law, and for three (3) years following the termination of this Agreement for Confidential Information that does not rise to the level of a trade secret. Upon any termination or expiration of this Agreement or as otherwise requested in writing by the disclosing Party, to the extent permitted by law, the receiving Party will return or destroy (as requested by the disclosing Party) all copies of the Confidential Information provided by the other Party and certify in writing that all copies of such Confidential Information have been returned or destroyed. Each Party shall be responsible for any use or disclosures by its employees or agents in violation of this Agreement. This Agreement and its Exhibits are not considered Confidential Information of either Party.
- Interference. EL will not intentionally cause radio frequency interference or physical interference with the operation of radio device(s) of third parties that are lawfully installed on City Street Lights. The City and PAID will incorporate the foregoing prohibition against interference in agreements with others than EL authorizing the use of Street Lights for the installation and operation of radio devices that use unlicensed spectrum and will enforce the prohibition to the extent permitted by applicable law, but, subject to applicable law, neither the City nor PAID shall have any other obligation with respect to the prevention of such radio frequency interference.
- 15 <u>Title and Quiet Enjoyment</u>. Subject to the rights of the City and PAID in this Agreement, and provided that EL is in compliance with the Agreement, EL shall have quiet enjoyment of the designated Street Lights for the term of the Agreement.
- 16 NO WARRANTIES. THE ONLY WARRANTIES BY EITHER PARTY CONCERNING THE SERVICES, NETWORK, SYSTEM, STREET LIGHTS AND OTHER EQUIPMENT AND PROPERTY OF THE CITY OR PAID, AND OTHER DELIVERABLES, OBLIGATIONS OR RIGHTS PROVIDED UNDER THIS AGREEMENT ARE THOSE EXPRESSLY CONTAINED HEREIN, AND ARE MADE IN LIEU OF ALL OTHER WARRANTIES AND REPRESENTATIONS, EXPRESS OR IMPLIED, INCLUDING, BUT NOT LIMITED TO, ANY

IMPLIED WARRANTIES OF MERCHANTABILITY AND FITNESS FOR A PARTICULAR PURPOSE OR OTHERWISE. EACH PARTY ALSO SPECIFICALLY DISCLAIMS ANY WARRANTIES OR REPRESENTATIONS RELATED TO THE INTERNET SECURITY-RELATED FEATURES OF THE SYSTEM AND/OR SERVICES. NEITHER PARTY CAN GUARANTEE SECURITY VIA THE INTERNET OR WIRELESS SERVICES AND NEITHER PARTY SHALL BE LIABLE FOR BREACHES OF SUCH SECURITY OR ANY OTHER RELATED WARRANTIES, EITHER EXPRESSED OR IMPLIED, INCLUDING WITHOUT LIMITATION, INTERNET FUNCTION AND/OR PERFORMANCE WARRANTIES. NEITHER PARTY SHALL HAVE LIABILITY TO THE OTHER PARTY FOR THE CONDUCT OF SUBSCRIBERS USING OR ACCESSING THE NETWORK, SYSTEM OR SERVICES. NOTWITHSTANDING ANY OF THE FOREGOING, THIS LIMITATION OF LIABILITY SHALL IN NO MANNER LIMIT OR WAIVE ANY RIGHT OR OBLIGATION OF EITHER PARTY UNDER SECTIONS 8 (INSURANCE; REMOVAL SECURITY) OR 9 (INDEMNIFICATION; INDEPENDENT CONTRACTOR; LIMITATION OF LIABILITY) OF THIS AGREEMENT.

17 Miscellaneous Provisions.

- 17.1 Contract Documents; Order of Precedence. The contract documents shall consist of the body of this Agreement (Sections 1-17) and Exhibits A-D hereto. In the event of conflict or variance between the body of this Agreement (Sections 1-17) and any other Agreement Document, the body of this Agreement (Sections 1-17) shall govern. Conflicts and variances among the contract documents shall be resolved by giving precedence in the following order: (i) The body of this Agreement (Sections 1-17); (ii) Exhibits A-D, which shall be of even precedence with each other.
- 17.2 <u>Business Privilege License</u>. EL shall obtain, and shall require its Subcontractors to obtain, a valid business privilege license (issued by the City's Department of Licenses and Inspections) as required by The Philadelphia Code.
- 17.3 Non-discrimination. In performing this Agreement, EL shall not discriminate or permit discrimination against any person because of race, color religion national origin, gender, or sexual orientation, and shall not discriminate or permit discrimination against any person who has AIDS or is HIV-positive. In the event of such discrimination, in material violation of the provisions of the Philadelphia Code referenced in this Section 17.3, the City may, in addition to any other rights or remedies available under this Agreement, at law or in equity, terminate the Agreement if EL has not complied with this Section 17.3 within thirty (30) days following written notice of a violation. EL agrees, in performing this Agreement, to comply with the provisions of the Fair Practices Ordinance of The Philadelphia Code (Chapter 9-1100). EL agrees to cooperate with the Commission on Human Relations of the City of Philadelphia in any manner, which the Commission deems reasonable and necessary for the Commission to carry out its responsibilities under Chapter 17-400 of The Philadelphia Code. Failure to so cooperate shall constitute a substantial breach of this Agreement entitling the City to all rights and remedies provided herein or otherwise available at law or equity.
- 17.4 <u>Chapter 17-1400</u>. EL shall comply with the terms and conditions set forth in Exhibit D, *Provisions Required by Chapter 17-1400 of the Philadelphia Code*, For purposes of Exhibit D, "Provider" means EL.

- 17.5 The Philadelphia Code, Chapter 17-400. In accordance with Chapter 17-400 of The Philadelphia Code, as it may be amended from time to time, EL agrees that its payment or reimbursement of membership fees or other expenses associated with participation by its employees in an exclusionary private organization, insofar as such participation confers an employment advantage or constitutes or results in discrimination with regard to hiring, tenure of employment, promotions, terms, privileges or conditions of employment on the basis of race, color, sex, sexual orientation, religion, national origin or ancestry, constitutes an event of default under Section 12, Default and Termination, of the Agreement entitling the City to all rights and remedies provided in the Agreement or otherwise available at law or equity. EL agrees to include the immediately preceding subsection, with appropriate adjustments for the identity of the Parties, in all subcontracts which are entered into for work to be performed pursuant to the Agreement. EL further agrees to cooperate with the Commission on Human Relations of the City of Philadelphia in any manner which the Commission deems reasonable and necessary for the Commission to carry out its responsibilities under Chapter 17-400 of The Philadelphia Code. Failure to so cooperate shall constitute an event of default under Section 12, Default and Termination. of the Agreement entitling the PAID, at the direction of the City, to all rights and remedies provided therein or otherwise available at law or equity.
- 17.6 Federal Laws. EL shall comply with the applicable provisions of Title VI of the Civil Rights Act of 1964 (42 U.S.C. Section 2000d et seq.), Section 504 of the Federal Rehabilitation Act of 1973 (29 U.S.C. Section 794), The Age Discrimination Act of 1975, (42 U.S.C. Section 6101 et seq.), Title IX of the Education Amendments of 1972, (20 U.S.C. Section 1681), and 45 C.F.R. Part 92, as they may be amended from time to time, which together prohibit discrimination on the basis of race, color, national origin, sex, handicap, age and religion.
- 17.7 Americans With Disabilities Act. EL understands and agrees that in the performance of this Agreement, no individual with a disability shall, on the basis of the disability, be excluded from participation in the Agreement or from activities or Services provided under the Agreement. As a condition of accepting and executing the Agreement, EL shall comply with all provisions of the Americans With Disabilities Act, 42 U.S.C. §§12101 et seq., and all regulations promulgated thereunder, as the Act and regulations may be amended from time to time, which are applicable (a) to Contractor, (b) to the benefits, Services, activities, facilities and programs provided in connection with the Agreement, (c) to the City, or the Commonwealth of Pennsylvania, and (d) to the benefits, services, activities, facilities and programs of the City or of the Commonwealth, and, if any funds under the Agreement are provided by the federal government, which are applicable to the federal government and its benefits, services, activities, facilities and programs. Without limiting the generality of the preceding sentence, EL shall comply with the "General Prohibitions Against Discrimination", 28 C.F.R. §35.130, and all other regulations promulgated under Title II of "The Americans With Disabilities Act", as they may be amended from time to time, which are applicable to the benefits, services, programs and activities provided by the City through contracts with outsider contractors.

17.8 The Philadelphia Code, Section 17-104. In accordance with Section 17-104 of The Philadelphia Code, EL by execution of this Agreement certifies and represents that unless EL has implemented the fair employment principles embodied in the MacBride Principles, (i) EL (including any parent company, subsidiary, exclusive distributor or company affiliated with Contractor) does not have, and will not have at any time during the term of this Agreement (including any extensions thereof), any investments, licenses, franchises, management agreements or operations in Northern Ireland and (ii) no product to be provided to the City under this Agreement will originate in Northern Ireland. In the performance of this Agreement, EL agrees that it will not utilize any suppliers, Subcontractors or subconsultants at any tier (i) who have (or whose parent, subsidiary, exclusive distributor or company affiliate have) any investments, licenses, franchises, management agreements or operations in Northern Ireland or (ii) who will provide products originating in Northern Ireland unless said supplier, subconsultant or Subcontractor has implemented the fair employment principles embodied in the MacBride Principles. EL further agrees to include the provisions of this Section 17.8, with appropriate adjustments for the identity of the parties, in all subcontracts and supply agreements which are entered into in connection with the performance of this Agreement. EL agrees to cooperate with the City's Director of Finance in any manner which the said Director deems reasonable and necessary to carry out the Director's responsibilities under Section 17-104 of The Philadelphia Code. EL expressly understands and agrees that any false certification or representation in connection with this Section 17.8 and/or any failure to comply with the provisions of this Section 17.8 shall constitute a substantial breach of this Agreement entitling the City to all rights and remedies provided in this Agreement or otherwise available in law (including, but not limited to, Section 17-104 of The Philadelphia Code) or equity. In addition, it is understood that false certification or representation is subject to prosecution under Title 18 Pa.C.S.A. Section 4904.

17.9 Executive Order 02-04: Gifts

- 17.9.1 Pursuant to Executive Order 02-04, no official or employee in the Executive and Administrative Branch of the City shall solicit or accept, directly or indirectly, anything of value, including any gift, gratuity, favor, entertainment or loan from any of the following sources:
- 17.9.2 Pursuant to Executive Order 02-04, no official or employee in the Executive and Administrative Branch of the City shall solicit or accept, directly or indirectly, anything of value, including any gift, gratuity, favor, entertainment or loan from any of the following sources:
 - a) A person seeking to obtain business from, or who has financial relations with the City;
 - b) A person whose operations or activities are regulated or inspected by any City agency;
 - c) A person engaged, either as principal or attorney, in proceedings before any City agency or in court proceedings in which the City is an adverse party;
 - d) A person seeking legislative or administrative action by the City; or
 - e) A person whose interests may be substantially affected by the performance or nonperformance of the official's or employee's official duties.

17.9.3 Pursuant to Executive Order 02-04, no official or employee in the Executive and Administrative Branch of the City is entitled to solicit or accept, and EL understands and agrees that if it offers anything of value to a City official or employee under circumstances where the receipt of such item would violate the provisions of this Executive Order, Provider shall be subject to sanctions with respect to future City contracts. Such sanctions may range from disqualification from participation in a particular contract to debarment, depending on the nature of the violation.

17.10 Certification of Non-Indebtedness.

EL hereby certifies and represents that EL, EL's parent company(ies), subsidiary(ies) and Subcontractors are not currently indebted to the City, and will not at any time during the term of this Agreement (including any extensions or renewals thereof) be indebted to the City, for or on account of any delinquent taxes (including, but not limited to, taxes collected by the City on behalf of the School District of Philadelphia), liens, judgments, fees or other debts for which no written agreement or payment plan satisfactory to the City has been established. In addition to any other rights or remedies available to the City at law or in equity, EL acknowledges that any breach or failure to conform to this certification may, at the option of the City, result in the withholding of payments otherwise due to EL and, if such breach or failure is not resolved to the City's satisfaction within a reasonable time frame specified by the City in writing, may result in the offset of any such indebtedness against said payments and/or the termination of this Agreement for default (in which case EL shall be liable for all excess costs and other damages resulting from the termination in accordance with Section 12, Default and Termination; provided, however, that if the breach or failure to conform is by a Subcontractor of EL (at any tier), the Agreement shall not be terminated if EL pays to the City the amount of the taxes, liens, judgments, fees or other debts owed to the City by the Subcontractor and constituting the basis for the breach or failure to conform.

- 17.11 <u>Compliance with Law and Regulations</u>. All work performed and services rendered shall strictly conform to all laws, statutes, and ordinances and the applicable rules, regulations, methods and procedures of all governmental bodies, boards, bureaus, office commissions and other agents.
- 17.12 Coordination of Public Statements. During the period prior to City Council's final approval of this Agreement, EL and PAID agree to meet regularly with the City and WP, or to talk by phone regularly, to coordinate the public statements that both will make during the City Council approval process. As part of this coordination the parties will exercise reasonable efforts to send a copy of press releases related to the Agreement and the Wireless Network to the other party before they are published to the extent there is sufficient time to comply with applicable law. Notwithstanding anything to the contrary, either party may make the statements it believes to be in its best interest or as is required to fulfill the obligations of that party or is required by law.
- 17.13 <u>Non-solicitation</u>. Neither party shall solicit the employment of any employee of the other party who has been assigned responsibilities under this Agreement, nor shall EL so solicit any such employee of the City, for the period of six (6) months following termination of

- responsibilities of each such employee; provided, that the prohibition in this Section 17.13 shall not apply to general advertisements of employment opportunities that are not directed to PAID or City employees.
- 17.14 <u>Prevailing Wages</u>. Contracts resulting from this Agreement shall require the payment of prevailing wages. Accordingly, employees of EL and its contractors performing work that would be subject to Section 17-107 of the Philadelphia Code, if performed for the City of Philadelphia, shall be paid at least the applicable prevailing wages for the respective occupational classifications designated in Section 17-107, in accordance with its requirements.
- 17.15 <u>Survival</u>. Any and all provisions set forth in the Agreement which, by its or their nature, would reasonably be expected to be performed after the termination of the Agreement shall survive and be enforceable after such termination, including, without limitation, the following:
- 17.15.1Any and all liabilities, actual or contingent, which shall have arisen in connection with the Agreement;
- 17.15.2EL's representations and covenants set forth herein;
- 17.15.3Those Sections referenced in Section 12.4.3.
- 17.15.4The following Sections:
 - Section 1, *Definitions*
 - Section 2.7, Conditions of the Obligations of the Parties
 - Sections 2.8 through 2.11, but only until all Communications Equipment is removed from all Street Lights
 - Section 2.14, *Compliance With Law*
 - Section 3.1, Construction Permits Required
 - The following subsections of Section 5. *Installation and Maintenance*, but only until all Communications Equipment is removed from all Street Lights in accordance as required under the Agreement:
 - **5**.1.2
 - **5.2-5.5**
 - **5.7**
 - **5.10**
 - Section 6, *Interruption to Service and Damage to Property*, but only until all Communications Equipment is removed from all Street Lights as required under the Agreement:
 - The following subsections of Section 7, *Compensation*:
 - 7.1 through 7.1.2
 - 7.2 through 7.4
 - Sections 8.1 through 8.4, *Insurance*, but provided that EL's insurance coverages are written on an "occurrence" basis, only until the earlier of (i) all Communications Equipment is removed from all Street Lights as required under this Agreement, or (ii) EL's letter of credit is required to be returned or cancelled under Section 8.5.; and if

- insurance coverages are not all written on an "occurrence" basis, then until the expiration of all applicable statutes of limitations on lawsuits arising under this Agreement.
- Section 8.5, *Removal Security*, but only until the earlier of (i) all Communications Equipment is removed from all Street Lights as required under this Agreement, or (ii) EL's letter of credit is required to be returned or cancelled under Section 8.5.
- Section 9, *Indemnification*; *Independent Contractors*; *Limitation of Liability*
- Section 10, *Dispute Escalation*
- Section 11.3, *Expiration and Removal*
- Section 12, Default and Termination
- Section 13, *Confidentiality*
- Section 16, *No Warranties*
- The following subsections of Section 17, Miscellaneous Provisions
 - 17.11, Compliance with Law and Regulations
 - 17.14, Prevailing Wages
 - 17.15, Survival
 - 17.16, Assignment
 - 17.17, *Notices*
 - 17.18, Governing Law
 - 17.19, Forum; Consent to Jurisdiction
 - 17.20, PAID's Expenses
 - 17.23, Third Party Beneficiaries
 - 17.25, *No Waiver*
 - 17.28, Agreement Drafted by All Parties
 - 17.30, Severability and Partial Invalidity
 - 17.31, Force Majeure
 - 17.32, Entire Agreement
- 17.16 Assignment. EL shall not assign this Agreement, or any portion of it, without the prior written permission of PAID and the City, which permission may be withheld in their complete discretion, and any such assignment made without such consent shall be void and shall not operate to relieve EL from any of its obligations or liabilities under this Agreement; provided, that EL is entitled to assign this Agreement without the consent of PAID and the City pursuant to the sale or transfer of all or substantially all of the assets or stock of EL, or pursuant to a transfer or assignment pursuant to a reorganization or merger or assignment to a subsidiary that is wholly or majority owned and controlled by EL; provided, that EL shall remain responsible for defaults or damages under the Agreement caused by such entity and for such entity's performance of the Agreement. PAID, with the City's approval but otherwise in PAID's sole discretion and upon written notice to EL, may assign the Agreement to another City-related authority or agency created pursuant to the Pennsylvania Municipal Authorities Act of 1945, as amended, or the Pennsylvania Economic Development Financing Law, Act No. 102, approved April 23, 1967, as amended, that has substantially the same powers, purposes, and authority as has PAID, including the power and authority to enter into and to carry out and perform this Agreement.

17.17 <u>Notices</u>. Any notice required or permitted to be given under the Agreement shall be given in writing and shall be personally delivered by hand with receipt obtained, by a national overnight express carrier (such as Federal Express), by facsimile (with copy by registered or certified United States mail) or by registered or certified United States mail, return receipt requested, addressed as follows:

IF TO PAID:

Philadelphia Authority for Industrial Development c/o Philadelphia Industrial Development Corporation Attn.: Samuel V. Rhoads, Sr. Vice President 2600 Centre Square West 1500 Market Street Philadelphia, PA 19102-2126

With copies to:

City Solicitor 1515 Arch Street, 17th Floor Philadelphia, PA 19102

IF TO EL:

EarthLink, Inc. 1375 Peachtree Street, Level A Atlanta, Georgia 30309 Attention: General Counsel Facsimile: 404-287-4905

With a copy to:

EarthLink, Inc.
1375 Peachtree Street, Level A
Atlanta, Georgia 30309
Attention: VP of Municipal Wireless Division
Facsimile: 404-287-1059

IF TO CITY:

Managing Director City of Philadelphia Municipal Services Building, Suite 1430 1401 JFK Boulevard Philadelphia PA 19102

With copies to:

Street Light Use Agreement EarthLink, Inc. Page 48 of 55 Commissioner Department of Streets Municipal Services Building, Suite 730 1401 JFK Boulevard Philadelphia PA 19102

and

City Solicitor 1515 Arch Street, 17th Floor Philadelphia, PA 19102

- 17.18 Governing Law. The Agreement and all disputes arising under the Agreement shall be governed, construed and decided in accordance with the laws of the Commonwealth of Pennsylvania.
- 17.19 Forum; Consent to Jurisdiction. The Parties agree that any lawsuit, action, claim, or legal proceeding involving, directly or indirectly, any matter arising out of or related to this Agreement, or the relationship created or evidenced thereby, shall be brought exclusively in the United States District Court for the Eastern District of Pennsylvania ("EDPA") or in a Pennsylvania state court sitting in Philadelphia County. It is the express intent of the Parties that jurisdiction over any lawsuit, action, claim, or legal proceeding shall lie exclusively in one of these forums. The Parties further agree not to raise any objection, as to forum or venue, to any lawsuit, action, claim, or legal proceeding which is brought in one of these forums, and the Parties expressly consent to the jurisdiction and venue of these forums. Nothing in this Section 17.19 shall be construed to waive the right of either Party to remove any action brought in a Pennsylvania state court to the EDPA if otherwise permitted by law.
- 17.20 <u>PAID's Expenses</u>. PAID shall have no obligation to take any action which it is permitted or required to take under this Agreement, whether at the direction of the City or EL, unless all reasonable costs and expenses of PAID, including its reasonable attorneys fees, are paid as incurred by the directing party.

17.21 Authority

17.21.1The City/PAID Street Light Use Agreement shall contain the following representation:

The City represents that the City has reasonably concluded that: (i) subject to the condition that City Council approves the City/PAID Street Light Use Agreement by ordinance and the ordinance becomes law, the City has the requisite power and authority to execute and deliver and to carry out and perform the City/PAID Street Light Use Agreement, and to grant to PAID the right to enter into and to carry out and perform the EL/PAID Street Light Use Agreement; (ii) the City has no actual knowledge of, nor has it received any legal notice of, any fact, act or omission which is inconsistent with this conclusion; and (iii) subject to the condition that City Council

approves the City/PAID Street Light Use Agreement by ordinance and the ordinance becomes law, its execution and delivery by the City is duly and validly authorized by all necessary action, and when executed and delivered, is the legal, valid, and binding obligation of the City, enforceable in accordance with its terms. The City hereby authorizes PAID to grant to EL the right to rely on this representation.

- 17.21.2PAID grants to El the same right to rely on the representation of the City provided in Section 17.21.1 that PAID has pursuant to the City/PAID Street Light Use Agreement.
- 17.21.3EL represents that (i) it has the requisite power and authority to execute and deliver and to carry out and perform the Street Light Use Agreement: (ii) it has no actual knowledge of, nor has it received any legal notice of, any fact, act or omission which is inconsistent with this conclusion; and (iii) its execution and delivery by EL is duly and validly authorized by all necessary action, and when executed and delivered, is the legal, valid, and binding obligation of EL, enforceable in accordance with its terms.
- 17.21.4EL grants to the City the same right to rely on the representation of EL provided in Section 17.21.3 that PAID has pursuant to this Agreement and hereby authorizes PAID to grant to the City the right to rely on such representation in the City/PAID Agreement.
- 17.22 Expiration of PAID. In the event that during the Term of this Agreement, the existence of the Authority is not renewed or extended pursuant to Applicable Law or the Authority otherwise ceases to exist, and a successor to the Authority with at least the same rights, obligations and sources of funding as the Authority hereunder is not established, all rights and obligations of the Authority under this Agreement shall automatically vest in the City without any further action by the parties hereto and the City shall be bound by all of the terms and conditions applicable to the Authority under this Agreement.

17.23 Third Party Beneficiaries.

- 17.23.1Provided that the City is bound by the terms and conditions of this Agreement as PAID is bound, El and PAID expressly agree that the City is an intended third party beneficiary of this Agreement and shall be entitled to enforce the terms and conditions of the Agreement (subject to and in accordance therewith) in the City's own name and on its own behalf against either or both of EL and PAID. Such rights of the City as third party beneficiary shall vest immediately upon execution of this Agreement and may not be modified or diminished except with the written consent of the City. Except as expressly provided otherwise in this Section 17.23 and elsewhere in the Agreement, nothing in the Agreement, express or implied, is intended or shall be construed to confer upon or give to any person, firm, corporation, or legal entity, other than EL, PAID and the City, any rights, remedies, or other benefits under or by reason of the Agreement.
- 17.23.2PAID and EL shall not breach, terminate or amend this Agreement in any way that would materially diminish or adversely and materially affect the grants to the City, the rights of the City, or the obligations of EarthLink or PAID to the City, that are set forth in this Agreement.

17.24 EL Third Party Beneficiary of City/PAID Agreement.

PAID agrees that the following provision shall be included in the City/PAID Street Light Use Agreement:

19 Third Party Beneficiaries.

19.1 PAID and the City expressly agree that EL is an intended third party beneficiary of the obligations of the City under the following Sections of this Agreement (collectively, the "EL Enforceable Provisions"): Section 2 (*Authorization; City Obligations*); Section 4 (*Term*); Sections 6.1 and 6.2 (*City Covenants*); Section 7 (*Representation of Authority*); Section 15 (*Survival*), but only as to the EL Enforceable Provisions that are set forth in Section 15; Section 17 (*Assignment*), but only as to clauses (ii) and (iii); Section 18 (*Notices*); this Section 19; and Sections 20.1-20.11 (*Miscellaneous*), but only to the extent directly related to and a condition to the enforcement of the provisions as to which EL has been expressly given third party beneficiary rights under this Section 19.1.

EL shall be entitled to enforce the terms and conditions of the EL Enforceable Provisions (subject to and in accordance therewith, and as to PAID and the City, subject to their respective limitations of liability set forth in the EL/PAID Agreement) in its own name and on its own behalf against either or both of the City and PAID, provided that in the exercise of such rights, (i) EL is bound by the terms and conditions of this Agreement as PAID and the City are bound thereby, provided, however, that the liability disclaimers provided with respect to PAID and the City in Section 16, *No Personal Liability*, shall not preclude EL's enforcement of the EL Enforceable Provisions against the City, and (ii) such rights may only be exercised in accordance with Section 19.2.

The foregoing rights of EL as third party beneficiary shall, subject to the conditions set forth in Section 19.2, vest immediately upon the Effective Date of this Agreement, and may not be modified or diminished except with the written consent of EL. Except as expressly provided otherwise in this Section 19 or elsewhere in this Agreement, nothing in this Agreement, express or implied, is intended or shall be construed to confer upon or give to any person, firm, corporation, or legal entity, other than PAID, the City and EL any rights, remedies, or other benefits under or by reason of the Agreement. PAID and the City agree that EL shall be entitled to exercise specific performance and all equitable remedies, except as provided in Section 19.3 below.

19.2 The rights of EL as third party beneficiary set forth in Section 19 shall be effective only if all of the conditions provided in Sections 19.2.1 and 19.2.2 are satisfied:

- 19.2.1 In the event that EL believes the City has by its act or omission failed to perform any of its duties or obligations as specified in the EL Enforceable Provisions or caused PAID to be in breach of the EL/PAID Agreement, then EL must certify the same to PAID and the City in writing and must assert in the certification EL's written demand that PAID enforce such duties or obligations of this Agreement against the City.
- 19.2.2 Upon receipt of such certification, PAID shall within twenty-five (25) days of such certification (or such earlier time as is reasonably specified in the certification if the certification reasonably states that irreparable damage is imminent) either (i) at EL's cost and using attorneys approved by EL, institute legal proceedings to enforce the City's duties or obligations as specified in the EL/PAID Agreement and in the EL Enforceable Provisions; or (ii) notify EL that it will not institute such legal proceedings. If PAID notifies EL that it will not institute such legal proceedings, or if, without such notification, PAID does not start such proceedings within the times specified above and/or does not then diligently and expeditiously pursue such proceedings, then the third party beneficiary rights of EL set forth in this Section 19 shall immediately be effective without further action on the part of PAID or EL and EL may thereafter institute such legal proceedings against the City in its own name.
- 19.3 Notwithstanding anything to the contrary in this Section 19 or elsewhere in this Agreement, or in the EL/PAID Agreement, EL shall not have the right under this Section 19 or otherwise under this Agreement to specific performance or any equitable remedy, including, without limitation, injunctive relief, against the City or PAID for the City's or PAID's failure to comply with Section 2.2 (*Authorization Not Exclusive*) of the EL/PAID Agreement or any provision of similar effect in this Agreement.
- 19.4 PAID and the City shall not breach, terminate or amend this Agreement or the EL/PAID Agreement in any way that would materially diminish or adversely and materially affect the grants to EarthLink, the rights of EarthLink, or the obligations of PAID to EarthLink, that are contained in the EL/PAID Agreement.
- 19.5 The City acknowledges that these provisions may operate for the benefit of EarthLink.
- 17.25 No Waiver. No delay or omission by either party hereto to exercise any right or power occurring upon any noncompliance or default by the other party with respect to any of the terms of this Agreement shall impair any such right or power or be construed to be a waiver thereof. A waiver by either of the parties hereto of any of the covenants, conditions, or agreements to be observed by the other shall not be construed to be a waiver of any succeeding breach thereof, or of any other covenant, condition, or agreement herein contained. Unless stated otherwise, all remedies provided for in this Agreement shall be cumulative and in addition to, and not in lieu of, any other remedies available to either party at law, in equity, or otherwise.

- 17.26 Notification. The City will with reasonable promptness notify EL of its receipt, subsequent to the execution of this Agreement, of applications by others for authorization to install a wireless internet access network on City Street Lights, where the circumstances of the applicant and the proposed wireless network are in material respects similar to the circumstances of EL and the Wireless Network. Notwithstanding Section 17-17, *Notices*, such notice may be delivered by City attorneys to EL attorneys, and may be given by electronic mail or telephone.
- 17.27 <u>Counterparts</u>. This Agreement may be executed in counterparts, each of which shall be on paper and each of which shall be deemed an original, and all of which shall constitute one instrument; provided, that the Agreement shall be effective and binding on the parties upon, but only upon, the execution by each party of one paper copy hereof.
- 17.28 <u>Agreement Drafted by All Parties</u>. This Agreement is the result of arms-length negotiations between the parties. The body of the Agreement (Sections 1-17) shall be deemed to be drafted by both parties and any ambiguities in the Agreement shall accordingly not be construed against either party.
- 17.29 <u>Headings, Table of Contents Not Controlling</u>. The table of contents and headings used in this Agreement are for reference and convenience only, do not in any way define, limit, describe or amplify the provisions of the Agreement or the scope or intent of its provisions, are not a part of the Agreement, and will not enter into its interpretation.
- 17.30 Severability and Partial Invalidity. Except as provided below, the provisions of this Agreement shall be severable; and if any provision of the Agreement, or the application thereof, for any reason or circumstance, is to any extent held to be invalid or unenforceable, the remaining provisions of the Agreement (as well as the application of all provision(s) that were held to be invalid or unenforceable to persons or entities other than those as to which they were held invalid or unenforceable) shall not be affected or impaired thereby; and each provision of the Agreement shall be valid and enforceable to the fullest extent permitted by law. Notwithstanding the foregoing, if the authorization to use Street Lights provided in this Agreement is terminated by PAID, or by order of a court of competent jurisdiction (following appeal and final disposition), or is prohibited by the enactment into law of a valid and enforceable federal, state, or local statute, all without EL's fault or EL's default of the Agreement, then EL shall have no obligation to pay to the City any dollar amounts provided in the Agreement that are due and owing to the City after the date of such termination.
- 17.31 Force Majeure. In the event that either party is unable to perform any of its obligations under this Agreement because of a Force Majeure event, as defined in this Section, the party that has been so affected shall immediately give notice to the other party; and shall exercise every commercially reasonable effort to resume performance of its obligations. Neither party shall be in default under Section 12, *Default and Termination*, if its failure to perform is the result of a Force Majeure Event and is without its fault or negligence, and the time for a party's performance shall be deemed extended by the time lost due to delay resulting from a Force Majeure Event. "Force Majeure Event" shall mean and include the following: acts of God, acts of third parties that are beyond a party's control, acts of

terrorism, strikes, civil riots, floods, material or labor restrictions by governmental authority, failures by providers of telecommunications service to the Wireless Network (other than EL), failures by providers of electricity to the Network, court order or change in applicable law prohibiting performance, or any other cause beyond a party's control.

17.32 Entire Agreement. This Agreement contains the entire agreement between the parties with respect to the subject matter hereof; supersedes all prior negotiations, agreements, and undertakings between the parties with respect to such subject matter; and cannot be changed, modified or amended except by agreement on paper and in writing signed by both parties. The parties hereto bind themselves, their heirs, executors, administrators, successors and assigns for the faithful performance of this Agreement.

IN WITNESS WHEREOF, the Parties grant, acknowledge and accept the terms, conditions and obligations of this Agreement as evidenced by the following signatures of their duly authorized representatives. It is the intent of the Parties that this Agreement shall become operative on the Effective Date.

EARTHLINK, INC.
By:
Name:
Title:
THE PHILADELPHIA AUTHORITY FOR INDUSTRIAL DEVELOPMENT
By:
Name:
Title: